

Invitation to Bid (ITB) ITB # VSW-TNK-2019-23

Department of Environmental Conservation Village Safe Water Program

Construction of 6-Inch Diameter Water Supply Well, Tununak, Alaska

Date of Release: February 5, 2019

Bid Opening Date: February 26, 2019 at 3:00PM (Alaska Time)

TABLE OF CONTENTS

Section	<u>Title</u>	<u>Pages</u>
1	Invitation to Bid	2
2	Instructions to Bidders	4
3	Award of Contract	2
4	Bid Form	1
5	Technical Specifications	26
6	Supplemental Contract Requirements	3
7	Attachments List	1

Section 1

Invitation to Bid

On behalf of the Native Village of Tununak IRA Council, the State of Alaska, Department of Environmental Conservation, Village Safe Water (VSW) Program is soliciting competitive sealed bids for the construction of 6-inch diameter water supply well. The VSW Program is the Native Village of Tununak IRA Council's authorized representative for this Invitation to Bid (ITB).

Bid Submittal

Sealed Bids must be submitted at the VSW Program Office, 555 Cordova Street (4th Floor), Anchorage, Alaska 99501 and must be time and date stamped prior to 3:00 PM Alaska Time, on **February 26, 2019**, at which time they will be publicly opened. Late Bids will not be considered.

Bidders must bid on all bid items in order to be considered responsive. The total price shall be quoted FOB Destination; Tununak, Alaska.

Background Information

Tununak is in Southwestern Alaska along the coast of the Bering Sea on the western edge of Nelson Island approximately 6 miles from the City of Toksook Bay. Tununak lies partially over Quaternary unconsolidated surficial deposits and folded Cretaceous sandstone and shale sedimentary rock. The Cretaceous sandstone and shale sedimentary rock underlie both Tununak and Toksook Bay along the western extent of Nelson Island.

Scope of Work

The work for the construction of 6-inch diameter water supply well portion of the project shall include the following in Tununak, Alaska:

- 1. Mobilizing, demobilizing, site cleanup and providing "tundra mats";
- 2. Drilling and installing 6-inch diameter steel cased water supply well;
- 3. Installing a 10-foot long well screen with a K-packer, gravel pack and blank casing;
- 4. Grouting of well casing;
- 5. Developing the well;
- 6. Test pumping of the well;
- 7. Temporary erosion control;
- 8. Installing well casing stick-up and a locking sanitary well cap;
- 9. Abandonment of well, if required.

The work completed consists of furnishing all labor, equipment, materials and for performing all work for the construction.

A standard workweek is based on a 60-hour workweek, Monday through Saturday, 7:30 a.m. to 6:00 p.m. Work in excess of this may cause community complaint of noise or dust, in which event work shall need to be accomplished within the above described 60-hour work window.

Minimum Qualifications of the Bidder

Reference Section 5, Technical Specifications, Paragraph 1.3, Minimum Qualifications.

Important Information to Bidders

In order to submit a responsive bid, the bidder shall include the written documentation specified below with their bid. Failure to include this documentation with the bid or documentation identified below shall cause the bid to be deemed non-responsive and therefore not considered.

- Complete the following:
 - a) Section 4, Bid Form;
 - b) Attachment A, Bid Schedule and Contractor's Questionnaire.

Pre-Bid Conference

A pre-bid conference will be held on February 14, 2019 at 09:00 AM Alaska Time, at the VSW Program Office, 555 Cordova Street (1st Floor), Anchorage, Alaska 99501, to provide specific project information, explain any unusual aspects of the project and address any potential bidder questions. Interested bidders are encouraged to attend the pre-bid conference or call the conference line number. The conference line number is 800-315-6338, access code 75021#. Pre-bid conference questions and answers will be issued as an addendum to the ITB by the Procurement Officer.

Questions regarding this solicitation shall be addressed in writing by email to the Procurement Officer, Christine Mash, at <u>DECDASPROCUREMENT@alaska.gov</u>. The deadline for submission of questions is 3:00 PM Alaska Time, **February 15, 2019.**

BIDDERS WITH DISABILITIES: The State of Alaska complies with Title II of the Americans with Disabilities Act of 1990. Individuals with disabilities who may need auxiliary aids, services, and/or special modifications to participate in this procurement should contact the Division of Administrative, Procurement Section at one of the following numbers no later than **February 11, 2019** to make any necessary arrangements.

Telephone: 907-269-0291 Fax: 907-269-7509

END OF SECTION 1.

Section 2

Instructions to Bidders

1. All sealed bids must be submitted prior to deadline as stated in the Invitation to Bid (ITB), Section One. Bids must be submitted in sealed envelopes. Oral, faxed or emailed Bids are not acceptable. Any bids received after the scheduled closing date and time will be deemed non-responsive and returned to the bidder unopened. Bids must be submitted in a sealed envelope marked:

From: Bidder Contract Name and Phone Number

Bidder's Address

ITB # VSW-TNK-2019-23

6-Inch Diameter Water Supply Well, Tununak, AK

To: DEC Village Safe Water (VSW) Program

Tununak Project

ITB # VSW-TNK-2019-23

Attn: Christine Mash

555 Cordova Street, 4th Floor Anchorage, Alaska 99501

It is the responsibility of the bidder to confirm receipt by the Procurement Officer identified in this ITB. Alternative bids for anything other than what is required will be rejected unless any possible alternatives are allowed in the ITB. By submitting a bid, the bidder agrees to the terms and conditions contained in this ITB, and that the VSW Program has the sole discretion to interpret and enforce the terms and conditions of this ITB. Award is contingent upon responsiveness and responsibility being determined at a later time.

All contractors must register with the Procurement Officer listed on the document to receive subsequent amendments. Registration entails providing the following information:

Company name, representative's name/title, email address and phone number.

When awarding the contract, VSW Program may choose to award the Base Bid plus all, any, or none of the Additive Alternates identified on the Bid Schedule. The determination of the apparent low bid will be based on the combined total of the Base bid and any awarded Additive Alternates.

Required Documentation from Bidders:

- Completed Bid Form (one page) Section 4
- Completed Bid Schedule and Contractor's Questionnaire (five pages) Attachment A

Bidders that fail to submit the required documentation, as identified above, before the deadline established for this ITB shall be deemed non-responsive.

2. Bids shall be submitted on the Bid Schedule and Contractor's Questionnaire provided as Attachment A and must be signed in ink. The person signing the Bid Schedule shall initial erasures or other changes appearing on the forms. Bid prices shall be in U.S. funds and include all applicable fees and costs to provide all items identified on the Bid Schedule.

3. A Bid may be withdrawn by an appropriate document duly executed in the same manner that a Bid must be executed and delivered to the place where Bids are to be submitted prior to the date and time for the opening of Bids. Upon receipt of such notice, the unopened Bid will be returned to the Bidder.

If a Bidder wishes to modify its Bid prior to Bid opening, Bidder must withdraw its initial Bid in the manner specified in the ITB and submit a new Bid prior to the date and time for the opening of Bids.

If within 24 hours after Bids are opened any Bidder files a duly signed written notice with the Procurement Officer and promptly thereafter demonstrates to the reasonable satisfaction of Procurement Officer that there was a material and substantial mistake in the preparation of its Bid, that Bidder may withdraw its Bid, and the Bid security will be returned. Thereafter, if the Work is rebid, that Bidder will be disqualified from further bidding on the Work.

- 4. Bidders shall acknowledge on the Bid Form their receipt of all ITB addendums issued during the bidding period. Failure to acknowledge addendums may result in the bid being rejected.
- 5. The bid must include everything necessary to fulfill all of the requirements of the ITB and other parts of the bidding documents. In the event of a difference between a price stated in words and a price stated in figures, the words shall govern. Math errors shall be corrected based on the criteria that words prevail over figures and that unit prices prevail over total prices (extensions of unit prices or bid summations). For purposes of bid comparison, no discounts will be considered.
- 6. The Procurement Officer reserves the right to reject any bid which is non-responsive, incomplete, or irregular; any bid which omits one or more items on the Bid Schedule; or any bid in which unit prices are unbalanced in the judgment of the VSW Program. The Procurement Officer reserves the right to waive any technicalities it deems appropriate in awarding the bid. The Procurement Officer reserves the right to reject any and all bids.
- 7. The VSW Program is not liable for any costs incurred by the bidder in bid preparation and submittal.
- 8. <u>BID GUARANTEE</u> is required with each bid in the amount of 5% of the amount bid. (Alternate bid items as well as supplemental bid items (if applicable) appearing on the bid schedule shall be included as part of the total amount bid when determining the amount of bid guaranty required for the project and shall reference VSW's project name and number.
- 9. Stantec Consulting Services, Inc. is the Engineer for this project.
- 10. <u>EQUIPMENT</u>: Equipment must meet or exceed the specifications listed on the Equipment Schedule. Equipment unit rates are hourly, with unlimited hours available. Only actual operating hours shall be paid for by the Owner. Operating hours are defined as hours that the equipment is in use and is being operated by an operator. Equipment that is idling must not be charged to the project. Equipment that is inoperable or not performing correctly, per the manufacturer, must be repaired or replaced within two calendar days, or as approved by the Engineer. Substituted equipment must meet or exceed the specifications listed on the Bid Schedule, and shall be mobilized/demobilized at the Contractor's expense. Equipment down time and standby time is not paid for. Equipment servicing and maintenance (labor, parts, and expendables) must occur after hours and shall be the responsibility of the Contractor, at no cost to the Owner. Fueling of the equipment must occur after hours and shall be the responsibility of the Contractor.

- 11. <u>SCHEDULE</u>: The estimated Period of Performance is March 2019 through September 1, 2019.
- 12. <u>CONSTRUCTION MANAGEMENT</u>: The work shall be directed by the Engineer and performed by the Contractor. Daily pay items shall be submitted by the Contractor daily on a form supplied by the Engineer. The form shall be reviewed by the Engineer on the next working day, and if approved, signed by the Engineer and Contractor. Items not identified will not be paid.
- 13. <u>BID BOND INFORMATION:</u> Reference Appendix A, Standard General Conditions of the Construction Contract, Article 11, Paragraph 11.3.
- 14. <u>PROTEST PROCEDURE</u>: Similar to AS 36.30.550 provides that an interested party may protest the content of the solicitation.

An interested party is defined in 2 AAC 12.990(a) (7) as "an actual or prospective bidder or offeror whose economic interest might be affected substantially and directly by the issuance of a contract solicitation, the award of a contract, or the failure to award a contract."

An interested party must first attempt to informally resolve the dispute with the procurement officer. If that attempt is unsuccessful, the interested party may submit a written protest. Written protest must include the following information:

- The name, address, and telephone number of the protester;
- The signature of the protester or the protester's representative;
- Identification of the contracting agency and the solicitation or contract at issue;
- A detailed statement of the legal and factual grounds of the protest including copies of relevant documents; and the form of relief requested.

All protests will be submitted to and responded to by the Procurement Officer IV as the protest decision authority. The appeal of a protest decision will be submitted to and responded to by the Procurement Officer IV and VSW Program Manager as the appeal decision authority. The appeal decision authority is the final decision and cannot be protested further. If protesting a solicitation document including the content of a specification, the protest must be filed with the Procurement Officer no later than **four** business days before quotations, bids, or proposals are due. Within **one** business day of receiving the protest, the Procurement Officer shall provide notice of the protest to all firms or persons that received the solicitation.

If protesting a decision to cancel a solicitation or the award of a purchase or contract, the protest shall be filed with the Procurement Officer within 10 calendar days of the date of the written Notice of Cancellation or Notice of Award. The deadline date cannot end on a weekend or state holiday. Within **one** business day of receiving the protest, the Procurement Officer shall provide notice of the protest to all firms or persons that received the solicitation and will acknowledge receipt of the protest. After protest receipt, the Procurement Officer shall take one of the following actions within 15 calendar days:

- a) Issue a written decision denying the protest including the specific reasons for the denial;
- b) Issue a written decision sustaining the protest in whole or in part and implementing an appropriate remedy.

If the protester is not satisfied with the protest decision, they may appeal the protest decision to the VSW Program Manager. The written appeal must be filed within 10 calendar days of the date of the protest decision. The deadline date cannot end on a weekend or state holiday. The appeal shall not raise any new issues that were not included in the written protest. An informal hearing on the protest appeal may be conducted by the VSW Program Manager to attempt to resolve the dispute. A written appeal decision on the appeal will be issued as follows:

- a) Issue a written decision denying the appeal; citing the specific reasons for the denial;
- b) Issue a written decision sustaining the appeal in whole or in part and implementing an appropriate remedy.

END OF SECTION 2.

Section 3

Award of Contract

- 1. Award will be made to the lowest responsive and responsible bidder, based on the total bid price in words on Attachment A.
- 2. For purposes of award, offers made in accordance with this Invitation to Bid (ITB) shall be good and firm for 90 days from the date of bid opening. If a contract is to be awarded, the Village Safe Water (VSW) Program shall issue a Notice of Intent to Award to the lowest responsive and responsible bidder as soon as practicable.
- 3. Pre Award Conference: After the Notice of Intent to Award, the apparent low bidder shall conference with VSW and the Engineer to discuss the project. VSW wants to confirm information in the bid and confirm that the bidder understands the project conditions. VSW Program will coordinate with the contractor to set up a date/time for the preconstruction conference.
- 4. Acceptance of the bidder's offer shall be by issuance of a construction services standard contract. The bidders total bid amount shall be construed as its offer, pursuant to the bid documents, to be accepted by the VSW Program. The terms and conditions of this ITB and bidders offer shall become part of any construction services standard contract resulting from this ITB. The contractor may not add additional or different terms and conditions to the ITB or any construction services standard contract resulting from this ITB.
- 5. If the successful bidder fails to enter into a contract then the obligation to the VSW Program created by the bid guarantee shall be in full force and effect. If the successful bidder enters into a contract, then the bid guarantee obligation shall be null and void.
- 6. Work shall commence after a notice to proceed is issued to the contractor by the VSW Program.
- 7. Prevailing Wages. The Contractor must comply with all requirements of Alaska Statute 36.05, entitled Public Contracts, Wage & Hour Administration, including the latest State of Alaska Department of Labor & Workforce Development Laborers and Mechanics Minimum Rates of Pay Pamphlet No. 600. http://www.labor.state.ak.us/lss/pamp600.htm is the website link for the current document. The Contractor is responsible for ensuring they use the most up-to- date version of Pamphlet No. 600.
- 8. The contractor shall furnish the following documentation to the Procurement Officer prior to contract award:
 - Valid Alaska Business License
 - Valid Alaska Contractor's License
 - Subcontractor List with required documentation (see requirements below)
 - Certificate of Insurance Appendix A, Article 11, Paragraph 11.2, General Requirements
 - Completed Federal Debarment Certification Form (two pages) **Appendix B**
 - Certification and Disclosure Regarding Payments to Influence Certain Federal Transactions
 Form (three pages) Appendix C

- 9. Subcontractors: Within five working days after the identification of the apparent low bidder for a construction contract, the apparent low bidder shall submit a list (**Appendix D**) of the subcontractors the bidder proposes to use in the performance of the construction contract. The list must include the name and location of the place of business for each subcontractor, evidence of each subcontractor's valid Alaska business license, and evidence of each subcontractor's registration as a contractor under AS 08.18. If a subcontractor on the list did not have a valid Alaska business license and a valid certificate of registration as a contractor under AS 08.18 at the time the bid was opened, the bidder may not use the subcontractor in the performance of the contract, and shall replace the subcontractor with a subcontractor who had a valid Alaska business license and a valid certificate of registration as a contractor under AS 08.18 at the time the bid was opened.
- 10. Project close-out occurs when the Contractor, Project Manager, the Review Engineer, the Inspectors, and the Procurement Agency complete all necessary paperwork, clearances, and documentation of the contract. Only issues of warranties and claims occur after this point.

END OF SECTION 3.

Section 4

Bid Form

ITB # VSW-TNK-2019-23 Construction of 6-Inch Diameter Water Supply Well, Tununak, Alaska

Company		
Address		
Do you qualify for MBE/WB] No. If yes, certification number: a documentation.
Го the Alaska Department of Envi	ironmental Conservation	ı, Village Safe Water Program:
In compliance with your Invitation andersigned proposes to provide the	` /	NK-2019-23 dated February 5, 2019 , the tified on the Bid Schedule.
contractor shall comply with all ap	oplicable federal, state, a	any contract resulting from this ITB, the and borough regulations, codes, and laws; bonds; and pay all applicable federal, state
The undersigned declares, under possibilitied was independently arrive		the laws of the United States, that the bid
The undersigned acknowledges recand date of each):	eipt of the following add	denda to the bid documents (give number
Addendum No	Date Issued	
Printed Name		Signature of Authorized Official
Title		Date
Phone Number		Email Address

Section 5

Technical Specifications

For

Construction of 6-Inch Diameter Water Supply Well, Tununak, Alaska

PART 1 - GENERAL

1.1 DESCRIPTION

The work covered by this section includes the furnishing of all materials, labor, equipment and performing all operations for the construction of a 6-inch diameter water supply well for the Native Village of Tununak IRA Council (herein referred to as OWNER) located in Tununak, Alaska. The Alaska Department of Environmental Conservation (ADEC) Village Safe Water (VSW) Program will act as the OWNER's Representative, assume all duties and responsibilities, and have the rights and authority assigned to OWNER in the Contract Documents in connection with completion of the work in accordance with the Contract Documents. Stantec Consulting Services, Inc. will serve as the Project Hydrogeologist/Engineer, hereinafter called ENGINEER, and will be responsible for monitoring the performance of services provided by the CONTRACTOR.

Attached to these specifications is Figure 1, which shows a vicinity map of Tununak, Alaska. Also attached to these specifications is Figure 2 that shows the two proposed water supply wells located adjacent to the airport material haul road in Tununak. The first water supply well to be drilled will be the well identified as Well #1 on Figure 1 and is located south of Unnamed Creek on the west side of the airport haul road. The second water supply well to be drilled, subject to the availability of sufficient funding as described herein, is identified as Well #2 on Figure 1 and is located north of Unnamed Creek on the west side of the airport haul road.

A hydrogeological desk top study of the Tununak area was completed in 2018 by the ENGINEER. The study evaluated known and assumed hydrogeologic conditions in Tununak and summarized their general findings as follows:

Tununak is in Southwestern Alaska along the coast of the Bering Sea on the western edge of Nelson Island approximately 6 miles from the City of Toksook Bay. Tununak lies partially over Quaternary unconsolidated surficial deposits and folded Cretaceous sandstone and shale sedimentary rock. The Cretaceous sandstone and shale sedimentary rock underlie both Tununak and Toksook Bay along the western extent of Nelson Island.

Currently, the deepest water well in Tununak is just over 30-feet deep. Based on the hydrogeological information on the existing water supply wells serving the City of Toksook Bay, it is assumed but not known that the depth of the proposed Tununak well may range from 100 to 200-feet below ground surface (bgs). The CONTRACTOR will be paid for a minimum depth of 100-feet even if the well's depth is less than 100-feet; however, the maximum well depth is not expected to exceed 200-feet in depth unless authorized by the OWNER and ENGINEER. Copies of well logs of water wells located in Tununak and the nearby Village of Toksook Bay are available upon request from the ADEC VSW.

The work covered by these specifications defines the furnishing of standard well drilling services in accordance with the general technical provisions of these specifications including the plant, labor, tools, equipment and materials and performing all operations in connections with the following work:

- Mobilizing, demobilizing, site cleanup and providing "tundra mats".
- Drilling and installing 6-inch diameter steel cased water supply well.
- Installing a 10-foot long well screen with a K-packer, gravel pack and blank casing.
- Grouting of well casing.
- Developing the well.
- Test pumping of the well.
- Temporary erosion control.
- Installing well casing stick-up and a locking sanitary well cap.
- Abandonment of well, if required.

Time is of the essence. All work shall be completed by **September 1, 2019** and be accomplished in accordance with these specifications and attached drawing. Time extensions can only be granted if there are changes in the work, or by excusable delays. No claim for extra cost shall be allowed to the CONTRACTOR for any delay created or within the control of the CONTRACTOR, including those delays caused by third parties, and beyond the control of the OWNER.

Pricing provided in the Bid Schedule is to include all costs associated with completing the project, including administration, overhead, profit, etc. Bidder must include a price for all items even if the price is \$0.00. Quantities referenced in the Bid Schedule are estimates only, and do not reflect the actual quantity that will be needed to complete the project. The OWNER does not guarantee a minimum or maximum quantity for any of these items. The CONTRACTOR shall invoice only for quantities used.

If the OWNER has provided optional items to select project work from, it will be the OWNER's right to select the option best suited for the desired product. The OWNER may elect to request services to drill a second well in the same general vicinity based on the bid unit prices subject to the availability of funding to complete the work. The Bid Schedule provides an Added Alternate bid item for additional well casing for the second well if it is determined a second well is needed and funding is available. No additional fees, cost, or monies will be owed to the bidder for options not selected.

Prior to award, upon the request of the OWNER, OWNER's Representative and ENGINEER, the Bidder shall attend a pre-award meeting with the OWNER, OWNER's Representative and ENGINEER, and shall bring to the meeting any documents requested by the OWNER, OWNER's Representative and ENGINEER to assist the OWNER in determining the Bidder's understanding of the Project. In the event the Bidder refuses to provide the requested information or attend the pre-award meeting, the OWNER may reject the bid as non-responsive.

1.2 REFERENCES AND APPLICABLE SPECIFICATION STANDARDS

The following publications of the most recent revised issues listed below that include regulations, standards and publications are hereby made part of this specification. These publications are referred to thereafter by basic designations only, form a part of this specification to the extent indicated by the reference:

- State of Alaska, Drinking Water Regulations 18 AAC 80.015 (Well Protection, Source Water Protection and Well Decommissioning) and 18 AAC 75 (Oil and Hazardous Substances Pollution Control).
- Alaska Best Management Practices for Construction of Non-Public Water Wells, Prepared by the Groundwater Protection Stakeholders Workgroup in Conjunction with Alaska Department of Environmental Conservation, Alaska Department of Natural Resources and the Alaska Water Well Association
- Alaska Best Management Practices for Maintaining or Decommissioning Water Wells and Boreholes, Prepared by the Groundwater Protection Stakeholders Workgroup in Conjunction with Alaska Department of Environmental Conservation, Alaska Department of Natural Resources and the Alaska Water Well Association.
- Environmental Protection Agency (EPA) Office of Water Supply, Manual of Water Well Construction Practices (formerly EPA-570/9-75-001).
- American Water Works Association (AWWA)
 - 200-75 Steel Water Pipe
 - 20167 Improving Well and Pump Efficiency
 - A100 Standard for Water Wells
 - C654-97 Standards for Disinfection of Wells
- American Welding Society
 - Section D3.0 Standard Welding Procedures
- American Public Health Association
- ASTM A-53 Specifications for Pipe, Steel, Black and Hot Dipped, Zinc Coated, Welded and Seamless
- ASTM A-139 Standard Specifications for Electric-Fusion (Arc) Welded Steel Pipe
- ISBN 0-9616456-001 Groundwater and Wells, Johnson Screens
- USCS Unified Soils Classification System

1.3 MIMIMUM QUALIFICATIONS

The well driller, herein referred to as CONTRACTOR, shall be a licensed general contractor with the State of Alaska in accordance with Title 8 (08.18.011) and be experienced and skilled in the drilling of water supply wells. The CONTRACTOR shall be qualified by broad experience in the construction

of municipal, community and residential drinking water wells, including the setting of screens, development of wells, and test pumping, and shall have the necessary equipment and experience to cope with all drilling problems that may arise. The CONTRACTOR shall employ competent workers for the execution of the well work and all such work shall be performed with modern equipment in good mechanical condition under the direct supervision of an experienced well driller. Information on the CONTRACTOR's experience shall be provided on the "Contractor Questionnaire" included in the Bid package with the understanding that the information shall be used for determination the CONTRACTOR meets minimum qualifications.

1.4 SUBMITTALS AND SUBSTITUTIONS

1.4.1 Materials, Methods and Schedule

- Prior to materials procurement or shipment, the CONTRACTOR shall identify all
 proposed products and materials required by the contract and provide manufacturers
 specifications for those products and materials.
- The CONTRACTOR shall identify all drilling equipment, methods and procedures proposed for completion of the project.
- The CONTRACTOR shall provide the proposed method of shipment and a tentative timetable adequate to meet the construction start date (to be determined prior to award of contract and project completion date of **September 1, 2019**.)

1.4.2 Safety and Protection

- The CONTRACTOR shall provide a Health and Safety Work Plan for review by the OWNER, OWNER's Representative and ENGINEER prior to award of contract.
- The CONTRACTOR shall be solely responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the WORK.
- The CONTRACTOR shall take all necessary precautions for the safety of, and shall provide the necessary protection to prevent damage, injury or loss to:
 - ✓ All employees on the WORK and other persons and organizations who may be affected thereby;
 - ✓ All the WORK and materials and equipment to be incorporated therein, whether in storage on or off the site; and
 - ✓ Other property at the site or adjacent thereto, including trees, shrubs, lawns, walks, pavements, roadways, structures, and utilities not designated for removal, relocation, or replacement during construction.
- The CONTRACTOR shall comply with all applicable Laws and Regulations (whether referred to herein or not) of any public body having jurisdiction for the safety of persons

or property or to protect them from damage, injury, or loss and shall erect and maintain all necessary safeguards for such safety and protection.

- The CONTRACTOR shall notify owners of adjacent property and utilities when prosecution of the WORK may affect them, and shall cooperate with them in the protection, removal, relocation, and replacement of their property.
- The CONTRACTOR shall designate a qualified and experienced safety representative at the site whose duties and responsibilities shall be the prevention of accidents and the maintaining and supervising of safety precautions and program.
- Materials that contain hazardous substances or mixtures may be required for the Work. A
 Material Safety Data Sheet shall be made available on site by the CONTRACTOR for
 every hazardous product used.
- Material usage shall be accomplished with strict adherence to Occupational Safety and Health Administration (OSHA) safety requirements and all manufacturer's warnings and application instructions listed on the Material Safety Data Sheet and on the product container label.
- The CONTRACTOR shall be responsible for coordinating any exchange of Material Safety Data Sheets or other hazard communication information required to be made available to or exchanged between or among employers at the site in accordance with Laws or Regulations.
- The CONTRACTOR shall notify the ENGINEER if it considers a specified product or its intended usage to be unsafe. This notification must be given to the ENGINEER prior to the product being ordered or, if provided by some other party, prior to the product being incorporated in the work.

1.4.3 Daily Drilling Report

- The CONTRACTOR shall complete and submit a daily drilling report to the ENGINEER at the end of each working day, either by fax or by electronic mail. The daily report requirement shall begin on the day the CONTRACTOR arrives on site and begins work.
- The report shall be submitted in an approved form as determined by the ENGINEER and include all applicable data to best describe work progress for that day.

1.4.4 Drilling Log (described in Section 3.3.2)

1.4.5 Well Log (described in Section 3.3.2)

1.4.6 Aquifer Test Pumping Log (described in Section 3.3.2)

1.4.7 Record of Plumbness and Alignment (described in Section 3.3.2)

1.4.8 Final Report (described in Section 3.3.2)

PART 2 – PRODUCTS

2.1 MATERIALS

2.1.1

Material used in the drilling process or well construction shall meet these requirements.

- All material used in the construction of the well shall be free of contaminants.
- All chemicals, substances and materials added to or brought in contact with water in a
 public water system well shall have either standard ANSI/NSF 60 or 61
 certification.
- All drill mud, additives, and lubricants shall have either standard ANSI/NSF 60 or 61
 certification. Drilling fluid or additives that contain biodegradable organic material,
 shall not be used during the drilling of a well.
- Potable water shall be used for drilling purposes. If necessary, the potable water shall be treated for drilling purposes in accordance with the drilling mud manufacturer recommendations. Potable water is available from the OWNER with the understanding the CONTRACTOR will be responsible for delivery of the water.

2.1.2

Well Casing shall meet the following requirements.

- The well casing used as a permanent part of the well structure shall be all new, seamless or electric resistance welded steel pipe conforming to ASTM-A53, Grade B, black wall. Casing shall be free of oil, grease or other contaminants;
- Shall have standard ANSI/NSF 61 certification for use in a public water system well;
- Shall be greater than minimum wall thickness and weight when required either to withstand the stresses of installation, grouting and operation, or corrosion;
- Shall be legibly marked on each length, by manufacturer, with all the following information:
 - i. Name of the manufacturer;
 - ii. Kind of pipe (continuous welded, electric resistance welded or seamless);

- iii. Weight or schedule;
- iv. Nominal or outside diameter;
- v. Specification number;
- vi. Heat or lot number;
- vii. Certification mark that verifies compliance with standard ANSI/NSF 61 for use in a public water system well.
- Shall be structurally sound and watertight throughout its length and shall have threaded and coupled or welded joints.

2.1.3 Drive Shoes

If permanent steel casing is to be installed by driving methods, the use of a manufactured drive shoe is required. The type and weight of the drive shoe shall be determined by the CONTRACTOR and specified in the initial materials submitted for approval. The shoe selected must be appropriate for exposing the screen by the pull-back method.

2.1.4 Well Screen

- The well screen shall be wire-wound, continuous slot. Screens shall be fabricated by circumferentially wrapping a triangular-shaped wire around a circular array of equally spaced internal rods. Each juncture between the horizontal wire and the vertical rods shall be fusion welded under water for maximum collapse strength. The wire shape must produce inlet slots with sharp outer edges, widening inwardly to minimize clogging. Screen end fittings shall be securely welded to each section.
- The well screen and its fittings shall be fabricated of the same material. This material shall be type 304 stainless steel unless otherwise specified. A manufacturer's certification of materials shall be provided to the purchaser.
- The well screen shall provide sufficient column and collapse strength to withstand installation and borehole pressures.
- Screen joints between screen sections and blank casing shall be welded or threaded and coupled.
- Packers for public water systems shall be constructed of materials that have standard ANSI/NSF 61 certification and use a rubber or neoprene seal.
- Screen shall be provided with a bail bottom or bottom plate fitting and a riser pipe.
- Where filter pack or formation stabilizer is installed, the screen shall have centralizers outside the top and bottom of the screen to ensure an even filter pack.

2.2 DRILLING

The well shall be of the tubular form. The CONTRACTOR shall select the method best suited to the kind and condition of material encountered, for obtaining the results in the most expeditious manner. Except the requirement that the minimum depth of the well is 100-feet, the OWNER makes no guarantee as to the linear footage that will be drilled. The CONTRACTOR will be paid for a minimum depth of 100-feet even if the well's depth is less than 100-feet; however, the maximum well depth is not expected to exceed 200-feet in depth unless authorized by the OWNER and ENGINEER.

2.3 CASING

Casing shall conform to the latest edition of the American Water Works Association Specifications for seamless steel pipe. No sub-standard or reclaimed material shall be used.

2.4 JOINTS

All well casing joints shall be made water tight, with welded joints or threaded couplings from above the ground surface to the bottom of the well, to exclude all surface drainage from entering the well. Welding shall be performed by a qualified welder in accordance with the Standard Welding Procedures of D3.0, of the American Welding Society (A.W.S.) Code. Welding rod shall meet the requirements of the A.W.S. for welding metal of the composition appropriate to the casing furnished by the CONTRACTOR.

2.5 WELL SCREEN

The CONTRACTOR is responsible to size the well screen with the approval of the ENGINEER. If needed, a well screen manufacturer can be contracted by the CONTRACTOR to provide recommendations as to screen length, design and placement in the aquifer. If a gravel (filter) pack is needed, the CONTRACTOR shall determine the recommended packing material to be used and appropriate screen slot size.

The CONTRACTOR shall furnish and install a well screen fitted with a self-sealing packer, including developing and pump testing the well to obtain accurate measurements on the potential productivity of the aquifer. The well screen shall be sized sufficient in length to provide well flow production requirements. The CONTRACTOR shall collect representative soil samples of the aquifer to be screened that may include one or more strata in the geologic formation. The CONTRACTOR shall engage the services of a recognized, competent soil testing laboratory to do gradation analysis of the soil samples. The gradation test results shall be used to select a recommended screen opening, length, and screen manufacturer. The size of the opening shall be determined in accordance with the effective size and uniformity coefficient of the soil found in the water bearing unit. The CONTRACTOR shall submit the gradation test results and recommended size of the well screen to the ENGINEER for review and consensus.

The CONTRACTOR shall furnish and install wire-wound stainless-steel well screen(s) with the rod and wire welded at each intersection, equivalent to Johnson Mild Steel, manufactured by Johnson

Division, U.O.P., Inc., St. Paul, Minnesota, or approved equivalent. The CONTRACTOR shall schedule the work so that a minimum amount of delay will be experienced in ordering and obtaining well screen(s). No additional payment shall be made because of delays or standby time in the ordering or procuring of well screen(s).

2.6 TEST PUMP

The CONTRACTOR shall furnish, install, and operate all necessary pumps, compressors, plungers, or other needed equipment including source of power and pump-top-waste piping that shall be capable of pumping at a minimum steady flow of 50 gallons per minute (gpm). The CONTRACTOR shall furnish all piping necessary to discharge water at a minimum of 25 feet from the pumping well. The CONTRACTOR shall also furnish, install, and maintain equipment of approved size and type for measuring the flow of water. The flow meter used shall be capable of instantaneous flow measurement and be equipped with a totalizer. Water level in the well during and after pumping shall be measured by an electrical probe or other equipment capable of measuring water level to the nearest 0.01 foot.

PART 3 – EXECUTION

3.1 GENERAL

It is anticipated that the well will be in an unconsolidated formation requiring casing throughout the entire depth. If consolidated formation is encountered, the need for full casing could be eliminated. The CONTRACTOR will consult with the OWNER's Representative and ENGINEER before proceeding. All work described for the completion of the well will be done in accordance with the requirements described herein as well as in accordance with the procedures outlined in the publication titled "Alaska Best Management Practices for Construction of Non-Public Water Wells" referenced above in Section 1.2.

The actual depth of the well will be that which is required to develop a suitable quantity of water, based on a minimum well yield of 25 gpm, sufficient to serve the OWNER's water supply requirements. In no case shall the CONTRACTOR exceed a 200-foot depth unless authorized by the OWNER and ENGINEER. When a satisfactory quantity and acceptable quality of water is encountered, subject to the ENGINEER's approval, the well will be developed, and pump tested in accordance with the specifications.

Based on limited hydrogeological information for the project area, it is anticipated the well will be drilled to a minimum depth of 100-feet bgs and possibly up to total depth of 200-feet; however, the completed well may have a shallower depth but not less than 50-feet subject to the discovery of an acceptable water bearing aquifer that meets the water supply needs of the OWNER. The CONTRACTOR will be paid for a minimum depth of 100-feet even if the well's depth is less than 100-feet; for bid solicitation purposes, a depth 200-feet for the completed well will be used. The entire length of the completed hole shall be cased including all consolidated and unconsolidated formations unless otherwise specified by the ENGINEER. The CONTRACTOR shall be prepared to address soil conditions including but not limited to heaving sands below the static water level. No additional payment will be allowed for difficult soil conditions.

3.2 MOBILIZATION AND DEMOBILIZATION

The CONTRACTOR shall be paid under a separate pay item to cover all costs for mobilization and demobilization of this project. These costs shall include all fixed costs anticipated by the CONTRACTOR to accomplish mobilization and demobilization for the project.

3.3 SAMPLES AND RECORD DOCUMENTS

3.3.1 Logging and Sampling During Drilling

The CONTRACTOR will maintain detailed records of the lithology and the water-bearing capacity of the strata penetrated during drilling. The CONTRACTOR shall collect and maintain formation samples at 5-foot depth intervals and each significant change in lithology. Sample containers to be furnished by the CONTRACTOR shall be heavy cotton bags, at least 8-ounce weight, or approved equivalent. A water proof label shall be attached to each sample

bag designating the well number, the exact top and bottom depth (measured to the nearest foot) at which the sample was taken, a description of the material, and how it was collected. The CONTRACTOR shall carefully note the depth at which ground water is first encountered and will estimate the relative amounts of ground water produced and the depths of productive zones to the nearest foot. The CONTRACTOR shall also record penetration rates as they pertain to the lithology of material penetrated. The CONTRACTOR shall note the depth at which bedrock materials are encountered, and the lithology of bedrock materials.

Selected water bearing formation samples shall be collected by the CONTRACTOR and submitted to a qualified third-party soil testing laboratory for particle size gradation analysis by ASTM Method D422 for determining optimal well screen slot size at the well site. The samples to be submitted for analysis will be collected after the total depth of the hole is reached. The CONTRACTOR shall ensure that each formation sample collected is of adequate volume for representative particle size analysis. All samples not tested for screen slot sizing shall be properly marked and kept at the well site for delivery to the OWNER.

3.3.2 Records

The grade, size, and length of the individual pieces of the casing and well screen shall be accurately measured, recorded, and reported. For each calendar day that equipment is at the site of the work, the CONTRACTOR shall submit a legible and neat daily report on the well drilling operations describing the drilling, the nature of the material encountered, the work accomplished that day, including items such as depths drilled, casing set, amount of sand removed during development work, the water level in the well at the beginning and end of each shift, the depth at which water was lost or found at any time during drilling, and such other pertinent data as required by the OWNER.

The CONTRACTOR shall complete the following forms to record the well construction data. These records shall be provided within thirty (30) days of completion of the well and delivered to the OWNER prior to final payment.

Two copies of the driller's log signed and dated by the well driller (or drilling machine operator) shall be given to the OWNER. One copy of the driller's log will also be provided to the ENGINEER, and one copy to the State of Alaska Department of Natural Resources Water Department. The log will record the materials penetrated to the nearest foot and other pertinent data. The CONTRACTOR shall complete a log of the well drilled under these specifications on the latest water well form provided by the State of Alaska Department of Natural Resources, Division of Mining, Land and Water. A form is available from the Alaska Department of Natural Resources (ADNR), http://dnr.alaska.gov/, 907-269-8400.

 All static water level measurements will be recorded, to the nearest 0.01 foot. This record, along with the times that the measurements were made, will be provided to the ENGINEER.

- A complete casing and screen location record will be made by the CONTRACTOR and given to the OWNER. This will show the lengths of each casing and screen section and the location of packers, plugs, and seal.
- Test pumping data as described in Section 3.7 from all pumping tests that were conducted on the well. Pump test records will show dates, location, water levels (to the nearest 0.01 foot), discharge rates, times of stopping and starting the pump, and other conditions that could affect the test data.

3.4 DRILLING AND WELL CASING

3.4.1 General

During drilling, no sand, dirt, rock, drill cuttings, drilling mud, or any foreign material whatever, other than potable water, shall be introduced into the well except with the prior knowledge and consent of the ENGINEER. The ENGINEER shall be entitled to make measurements at any time with his own equipment at no extra cost for standby time, provided actual drilling or pumping time lost does not exceed 30 minutes per event.

3.4.2 Plumbness and Alignment

3.4.2.1.1 General

The completed well shall be constructed round, plumb, and true to line, as defined in this standard. Tests for plumbness and alignment shall be made after completed construction of the well; however, the CONTRACTOR may make during the work whatever tests he may deem necessary for his control of the work.

3.4.2.1.2 Tolerances

The tolerances described below shall be maintained by the CONTRACTOR from the top of the well casing to the top of the well screen.

Plumbness: The maximum allowable horizontal deviation (drift) of the well from the vertical shall not exceed two-thirds of the smallest inside diameter of that part of the well being tested per 100 feet (30.5 meters) of depth.

Alignment: A pump of nominal diameter 2 inches smaller than the nominal casing size shall be able to be installed in the completed well to the top of the well screen without bending, as outlined in the procedures given in AWWA Standard A100.

Correction of Faulty Plumbness and/or Alignment: Should the completed well fail to meet the tolerances specified above, the CONTRACTOR shall, at his own expense, correct such defects. As an alternative to correction of the defects, the OWNER may offer to negotiate a reduction in the contract price in proportion to the

damage which would be experienced due to decreased production potential or increased operation costs if the well was accepted without correction of said defects.

If the CONTRACTOR fails to complete the well according to the specifications, the hole shall be declared abandoned. If the CONTRACTOR voluntarily stops work and/or fails to complete the well in a satisfactory manner within the overall time limit set be the contract, or as may be extended under the terms of the contract, it shall be considered abandoned.

No payment will be made by the OWNER for any part of the drilling or pumping of, or the casing or screen used in an abandoned hole. The CONTRACTOR may salvage the materials installed in an abandoned hole. The CONTRACTOR is reminded that any hole which is abandoned must be sealed, protected, or filled as required to secure the well using a method approved by the ADEC.

3.4.3 Drilling

General: The CONTRACTOR shall coordinate and continually inform the ENGINEER of drilling progress and stratigraphic information and shall not drill beyond the depth authorized by the ENGINEER. A description and specifications of any drilling or development fluids or additives which the CONTRACTOR may desire to use during the work shall be submitted to the ENGINEER in writing and approved for use prior to being used on the project.

The drilling may be accomplished by a cable tool, a conventional hydraulic (mud) rotary or an air rotary drilling machine. The equipment used shall be in good repair and be capable of drilling, as specified herein, to a depth of 300 feet. Rotary drilling machines shall be equipped with a sand line and sand pump bailer.

The drill rig must be a 4-wheel drive vehicle (truck mounted) capable of traversing tundra terrain typical in the Tununak area as found in the immediate vicinity of the proposed well sites. The well sites are located approximately 50-feet from the edge of the airport haul road. Photographs of the proposed drill sites are available upon request from the ADEC VSW. In addition, the CONTRACTOR shall provide "tundra mats" to accommodate the overland movement of the drill rig to minimize damage to the underlying tundra. The "tundra mats" must be at least 60-feet in length.

The CONTRACTOR shall drill a 6-inch diameter water well to accommodate the 6-inch well casing without over boring that would produce loose casing. The hole shall be drilled to a productive zone or up to 200 feet. If a satisfactory water bearing formation is found at a lesser depth, the drilling shall be terminated and the well completed to that depth with the approval of ENGINEER. When drilling in a potential aquifer and conventional rotary drilling is used, the CONTRACTOR may be required by the ENGINEER to periodically stop drilling, remove any material from the hole and check for yield using either compressed air or a bailer. This checking shall be considered a normal part of the drilling procedure and no additional payment will be made for it.

When sufficient yield is indicated from a water bearing stratum the CONTRACTOR shall, after obtaining the approval of the ENGINEER, install the well screen and develop the well according to the provisions herein. A minimum water well yield of 25 gallons per minute is being sought. If a stratum, which will produce in excess of 25 gpm, is penetrated, the water well shall be developed to produce the maximum yield obtainable through the appropriate well screen.

Cable-Tool Method: The selection of drill bits, bailers, associated tools to be used, and their application to the drilling task shall be so done as to maximize penetration rate and minimize caving of the hole to a diameter greater than the casing outside diameter (O.D.). To improve the accuracy of samples collected from water bearing formations, grinding of the aquifer materials shall be minimized by using only sufficient drilling to loosen the material for removal by the sand pump. To prevent "mudding off" of the water bearing formations, sufficient water shall be bailed from the bottom of the hole to minimize concentration of fines in the pore spaces of the formations. If heaving formations are encountered, however, it may be necessary to curtail water bailing to prevent excessive heaving.

Air Rotary Method: Procedures used in the air rotary drilling method shall be chosen to minimize heaving and caving of the formation materials into the hole. The addition of approved foaming agents to the air stream injected down the hole may be helpful in maintaining the stability of the hole. If, however, heaving cannot be controlled, especially when drilling through potential production zones, the CONTRACTOR shall be required to discontinue use of the air rotary method. In that case, advancement of the hole may only be continued by means which will prevent heaving, such as sand pumping and driving using the sand line and casing driver on the rig. If that proves unworkable, substitution of a cable tool rig may be necessary.

Reverse Circulation Rotary Method: Drilling of the hole shall not begin until the well screen, casing, gravel pack, and all other required materials and equipment have been delivered to the site and are fully prepared for the work to begin. In case insurmountable hole stability or lost circulation problems occur, the CONTRACTOR shall be prepared to stabilize the hole by filling (partially or completely) with the gravel pack material, running casing in the hole and completing the drilling by the cable tool or air rotary method. Since such an emergency action would be substantially altering the original well design, the CONTRACTOR must coordinate the various details of such a plan with the ENGINEER.

3.4.4 Well Casing

3.4.4.1 General

This section sets forth standards applicable to permanent casings for water wells. Selection of temporary casings used only for construction is left to the CONTRACTOR unless otherwise specified herein. Permanent well casings shall be continuous and watertight from the top to the bottom of well, except for screened

areas, or open hole completions in bedrock as approved by the ENGINEER. The finished top of casing (stick-up) shall be a minimum of twenty-four (24) inches above ground level and shall have a locking vandal-proof sanitary seal (well cap).

3.4.4.2 Well Casing Diameter

The minimum size casing to be installed in the well shall be nominal 6-inch inside diameter as called out by the Bid Schedule. Unless provided for in the Bid Schedule, no consideration will be made for any over-size casings, which the CONTRACTOR may deem necessary to reach the specified depth.

3.4.4.3 Well Casing Joints

Casing joints may be either welded or threaded, at the CONTRACTOR's option. Standards for jointing shall be in accordance with AWWA Standard C206.

3.4.4.4 Drive Shoe

A drive shoe shall be installed on the bottom end of each string of casing in the well, which must be driven into position. The drive shoe may be omitted from a casing string, which is lowered into a completed hole drilled by the reverse circulation method.

3.4.4.5 Well Casing Installation

Cable Tool or Air Rotary Method: Except where local experience confirms that a stable open hole may be drilled (in permafrost or poorly consolidated sediments), the casing shall be driven into the hole as required to follow the bit closely. The contour of the drive head shall be checked periodically and re-faced as necessary to prevent deformation of the pipe end. In consolidated rock formations, the casing shall be seated by driving it into the surface of the rock until a seal is obtained. At the CONTRACTOR's option, he may use the pull-down method to augment or replace the driving effort required to advance the casing. The CONTRACTOR is cautioned, however, that regardless of the method used to advance the casing, care shall be exercised to avoid damage to the casing pipe and/or joints of the permanent casing string.

Reverse Circulation Method: When the reverse circulation drilling method is used, the casing assembly shall be lowered into the hole as quickly as possible after total depth has been reached and the hole is stabilized (caving and/or excessive water loss has been controlled).

3.5 WELL SCREEN

3.5.1 General

The CONTRACTOR shall be prepared to furnish and install the specified well screen of nominal diameter and general length as indicated by the Bid Schedule. Slot size and exact total length shall be determined by the CONTRACTOR subject to the ENGINEER's acceptance from test data and material samples taken from the aquifer.

3.5.2 Screen Design and Delivery

The CONTRACTOR shall be responsible for the design and delivery of the well screen. Well screens shall be required in unconsolidated formations. The well screen shall be equipped with a packer that is self-sealing neoprene with a mild steel or stainless-steel collar ("K" packer). For bid award purposes, a length of 10-feet for the well screen will be used. A minimum 5-foot riser pipe shall be welded to the top of the screen and a 2-foot long extension pipe welded to the bottom of the well screen, both pipes to be same diameter of the screen. The bottom extension pipe shall be closed off with a welded mild steel fitting.

This period for screen design and delivery shall be considered as incidental to the contract and shall not be basis for a claim for contract time extension. Neither shall the time required for delivery of the screen be allowed as the basis for a time extension claim.

3.5.3 Pull-Back Method

Basis of Design: If the well is drilled by the drill-and-drive/pull-back method, either cable tool or air rotary the screen design shall be based on samples collected from the cuttings removed from the well.

Screen Installation (Pull-Back Method): The CONTRACTOR shall adhere to the following procedure for installing the well screen:

- Backfill the inside of the casing with clean gravel to the design depth of the bottom of the well screen assembly.
- Pull the casing back to the elevation of the bottom of the well screen assembly as directed by the ENGINEER or cut the casing at that level.
- Place the well screen in proper position.
- Pull the casing back to provide the amount of overlap at the top of the screen as required by the ENGINEER.
- If a lead packer is used, expand the packer against the inside of the well casing using a wedge to provide a sand-tight seal above the well screen.

3.5.4 Open Hole Method

Basis of Design. If a well is drilled by the reverse circulation method, the hole size drilled shall be large enough to allow installation of an artificial gravel pack of at least a 4-inch thickness on all sides of the well screen. The details of gravel pack gradation and depth, as well as screen slot size and location shall be determined by the ENGINEER from data collected from each well.

Screen Installation (Open Hole Method). Pipe-sized screen may be welded into the casing string at the locations as directed by the ENGINEER and lowered into the open hole as a continuous assembly. After the casing is lowered to the prescribed level, the gravel pack material shall be placed by a tremie pipe to prevent its segregation. The screen shall be centered in the hole. Since the well screen has limited column strength, the CONTRACTOR shall use care to prevent deforming the screen during its installation. The installed screen shall be sufficiently round and straight to allow the passage of a 15-foot sand pump or bailer of 2 inches smaller O.D. than the I.D. of the screen throughout its length.

As an option, the CONTRACTOR may choose to use a telescope size screen and follow the pull-back method as covered in Section 3.5.3. If this procedure is chosen, the bottom 30 feet of the casing shall be centralized at 10-foot intervals. Also, the screen assembly and gravel pack shall be in place before the casing is pulled back. This method has the advantage of minimizing the chance of deforming the screen during its installation.

The gravel pack shall extend approximately 30 feet above the highest screen in the well, with the remainder of the annulus filled with grout. See Section 3.13, Well Grouting. A 2-inch galvanized steel pipe shall be left in the annulus to the top of the gravel pack through which additional gravel may be added as required.

Disinfection of Materials. Gravel backfill, and gravel pack material used in well screen installation shall be disinfected by maintaining a chlorine residual in the water in the well of at least 50 mg/l during the placement of those materials.

3.6 WELL DEVELOPMENT

Well development shall consist of the following:

- The water bearing horizon around the screen shall be thoroughly surged and developed using techniques approved by the screen manufacturer. The CONTRACTOR shall notify the ENGINEER at least 24 hours prior to conducting well development. The CONTRACTOR shall keep a log of development activity and results. The CONTRACTOR shall develop the well to the desired production yield, with a minimum of 25 gpm, if possible, using care in the performance of the work to prevent breakdown or caving-in of strata overlying that from which water is to be drawn. The screen must not have changed alignment after installation.
- The CONTRACTOR shall develop, pump, or bail the well by such methods as may be approved
 by the ENGINEER until the water pumped from the well is free from sand and until the turbidity

is less than 1 NTU as described in *Standard Methods of the Examination of Water and Wastewater for the Nephelometric Method*. The methods of well development, including aquifer stimulation (hydrofracturing) shall meet the procedures outlined in the publication titled "Alaska Best Management Practices for Construction of Non-Public Water Wells" referenced above in Section 1.2.

- Since it is impossible to predict the number of hours that may be required to properly develop the
 well, the number of hours given in the Bid Schedule is an estimate only and the actual number will
 be determined by the results as development proceeds.
- Upon completion of development and sand extraction work, the well shall be tested by the CONTRACTOR in accordance with the Sand Content Test, as described in Article 52.000-001-000 of the EPA *Manual of Water Well Construction Practices*. The sand content shall be determined by averaging the results of five (5) samples collected at the following times during the final pumping test: 1) 15 minutes after start of the test; 2) after ½ of the total planned test time has elapsed; 3) after ½ of the total planned test time has elapsed; and 5) near the end of the pumping test. A "Rossum" Sand Tester (Roscoe Moss Company of Los Angeles) or equivalent approved by the ENGINEER shall be used for the Sand Content Test.

The minimum volume of water sample collected for testing for sand content shall be the test rate of flow in gallons per minute (gpm) multiplied by 0.05, with the exception that the minimum required for wells tested at less than 100 gpm shall be 5 gallons (U.S.).

3.7 TEST PUMPING

3.7.1 General

The CONTRACTOR shall furnish, install, and operate all necessary pumps, compressors, plungers, or other needed equipment (including source of power and pump-top-waste piping) that shall be necessary to determine the average yield of water per foot of drawdown. The test pump shall be capable of pumping at a rate of 50 gpm, if possible. The CONTRACTOR shall also furnish, install, and maintain equipment of approved size and type for measuring the flow of water. The flow meter used shall be capable of instantaneous flow measurement and be equipped with a totalizer.

The CONTRACTOR shall provide all necessary pump, piping, hoses, and other equipment, including power, to test each well. Payment for test pumping as specified shall be made at the unit price bid based on the number of hours that the test pump is operated and the number of hours that recovery measurements are being taken. No payment shall be made for time spent setting and removing the test pump, or for an incomplete pump test because of pump failure. Pump test water shall be discharged a minimum of 25 feet from the pumping well.

3.7.2 Testing

After the well has been completely constructed, developed, cleaned out, and the depth accurately measured, the CONTRACTOR shall notify the ENGINEER and then shall make ITB # VSW-TNK-2019-23

all necessary arrangements for conducting the final yield and drawdown test. Yield and drawdown tests shall include a "constant rate" test as outlined below.

Constant Rate Test: The CONTRACTOR shall commence constant-rate testing at a rate determined by the ENGINEER. The cost for this recovery period will be considered incidental to the hourly pumping and no additional payment will be made to the CONTRACTOR. During the testing and recovery period, the CONTRACTOR shall be responsible for water level measurement in the well using an electric probe capable of measuring water level to the nearest 0.01 foot. The frequency of required water level measurements is listed below.

Measurement Frequency in Pumping Well During Pumping and Recovery

Time Period Since Test Began (minutes)	Frequency of Water Level Measurement and Totalizer Readings
Immediately prior to testing	Three successive water level measurements within 0.03 feet.
0 - 10	0.5 minute
11 – 15	1 minute
16 – 60	5 minutes
61 – termination of test (maximum 24 hours)	30 minutes

The frequency of measurements shall revert to 1-minute intervals for a 10-minute period after each change in discharge rate.

Test pumping shall be accomplished under steady flow conditions at a rate as determined by the ENGINEER. Continuous and accurate measurement of discharge rates shall be maintained, and any changes of rate noted in the records. The water shall be measured, and the depth recorded on a scheduled basis to provide an accurate account of water drawdown, pumping rates, and time. Water level in the well casing shall be measured with an electrical water level detector. A ¾-inch diameter PVC pipe shall be installed in the well with the test pump to facilitate operation of the detector. Test pumping shall be continuous and without interruption until data is obtained to determine well yield; however, the CONTRACTOR will not be required to maintain more than 24 hours of continuous pumping of the well. No payment for test pumping will be made for pumping time performed in which the operation of pumping and measurement equipment does not meet these requirements.

It is imperative that data collected from pump testing be complete and accurate. Payment will not be made for a "failed" test. For the purposes of this project, a pump test is considered failed if one or more of the following occurs:

- Failure to record water level measurements in the approximate time intervals shown above, with an accuracy of at least 0.10 feet.
- Failure to maintain a constant pumping rate within +/- 15 percent.
- Failure to record pumping rates and totalizer readings at the approximate intervals shown above.
- Failure of the pump for a period of 10% or greater of the lapsed time of pumping.

If a failed test occurs, the CONTRACTOR shall be required to remedy the problem and allow recovery of the well for a duration of two times the pumping duration before re-starting the test. The test may be restarted earlier if the water level in the well equilibrates as determined by the ENGINEER.

After completion of the constant-rate test, the CONTRACTOR shall monitor recovery in the well for the same duration as the test, and at the approximate time intervals shown for the pump test. The cost for the recovery monitoring shall be incidental to the hourly pumping cost. No additional payment will be made for recovery monitoring. No disturbance to the well that would affect water levels (such as removal of the pump) shall be made by the CONTRACTOR during the recovery period.

After recovery and temporary pump removal, the CONTRACTOR shall remove any sand or other objectionable material that may have accumulated in the well during testing operations and re-measure the depth of the well. The water pumped from the well shall be disposed of as approved by the ENGINEER. It will be the responsibility of the CONTRACTOR to dispose of all water in a manner that will not create a hazard or nuisance to the nearby roads, buildings, or other property.

3.8 DISINFECTION

3.8.1 Scheduling Disinfection

The CONTRACTOR shall provide for disinfection as soon as construction of the well and cleaning procedures have been completed. The CONTRACTOR shall carry out adequate cleaning procedures immediately preceding disinfection where evidence indicates that normal well construction and development work have not adequately cleaned the well. All oil, grease, soil, and other materials, which could harbor and protect bacteria from disinfectants, shall be removed from the well. Unless prior approval is obtained for employing chemicals or unusual cleaning methods, the cleaning operation is to be carried out by pumping and swabbing only. Where test pumping equipment is to be utilized, such equipment shall be installed prior to or during disinfection and be thoroughly hosed, scrubbed, or otherwise cleaned of foreign material.

3.8.2 Disinfectants

Chlorine shall be used as a disinfectant. The disinfectant shall be delivered to the site of the work in original, closed containers bearing the original label indicating the percentage of available chlorine. The disinfectant shall be recently purchased. Chlorine compounds in dry form shall not be stored for more than one year, and storage of liquid compounds shall not exceed 60 days. During storage, disinfectants shall not be exposed to the atmosphere or to direct sunlight. Unless superseded by government regulations, the quantity of chlorine compounds used for disinfection shall be sufficient to produce a minimum of 100 parts per million available chlorine in solution when mixed with the total volume of water in the well.

3.8.3 Procedure

Unless otherwise modified, due to problems involved with a specific well or conflict with local, State or Federal governmental regulatory agency requirements, the disinfection procedure shall include, but not be limited to, the following criterion:

- Provide reliable means for insuring that the disinfecting agent is uniformly applied
 throughout the entire water depth of the well without relying on subsequent mechanical
 or surging action for dispersing the disinfectant; the dispersion of the disinfectant shall be
 assisted by pouring into the well a volume of water equal to the volume of the screen, after
 the disinfectant has been emplaced. This will cause the disinfectant to flow out of the well
 into the area adjacent to the screen.
- All accessible portions of the well above the water level shall be maintained in a damp condition with water containing the required concentration of disinfecting agent for a period of not less than 20 minutes.
- The disinfecting agent shall be left in the well for a period of at least 12 hours. After a 12-hour, or longer contact period the well shall be pumped to clear it of the disinfecting agent. The disposal point for the purged water shall be selected by the CONTRACTOR and approved by the OWNER and ENGINEER to minimize potential damage to aquatic life or vegetation.

3.9 WATER QUALITY TESTING

The CONTRACTOR shall notify the ENGINEER three working days in advance of anticipated pump testing of the well. The ENGINEER will be present to collect representative water samples in laboratory supplied containers with the necessary preservatives. The water sample for laboratory analysis shall be collected just prior to the completion of the constant rate pump test. During drilling and test pumping of the well, the ENGINEER may elect to collect additional water samples for onsite field-testing purposes to evaluate the general water quality characteristics of the well water. Water quality tests will be paid by the OWNER.

3.10 PROTECTION OF QUALITY OF WATER

The CONTRACTOR shall take such precautions as may be required to permanently prevent contaminated water, or water having undesirable physical or chemical characteristics, from entering through the opening made in drilling the well or the stratum from which the well is to draw its supply. The CONTRACTOR shall also take all necessary precautions during the construction period to prevent contaminated water, gasoline, etc., from entering the well either through the opening or by seepage through the ground surface. In the event the well becomes contaminated, or water having undesirable physical or chemical characteristics enters the well due to the neglect of the CONTRACTOR, the CONTRACTOR shall, at his own expense, perform such work to supply such casing, seals, sterilizing agents or other materials as may be necessary to eliminate the contamination or to shut off the undesirable water.

3.11 TEMPORARY EROSION AND POLLUTION CONTROL

Temporary erosion and pollution control shall consist of the following:

- The CONTRACTOR shall provide temporary control measures to control temporary erosion and water pollution in accordance with local, state, and federal regulations or ordinances.
- Prior to the start of the applicable construction, the CONTRACTOR shall submit to the ENGINEER for acceptance, the CONTRACTOR's plan for accomplishment of temporary erosion control work, including disposal and control of water discharged during well development and pump testing. No work shall be started until the erosion control plan and proposed methods of operations have been accepted.
- This work shall involve the construction of temporary berms, dikes, dams, sediment basins, and slope drains, and the use of temporary mulches, seeding, or other control devices or methods as necessary to control erosion. Temporary pollution control measures shall be used to correct conditions that develop during construction that were not foreseen during the design stage.
- In the event of conflict between these requirements and pollution control laws, rules, or regulations of other federal, state, or local agencies, the more restrictive laws, rules, or regulations shall apply.
- Other miscellaneous tasks described in this section as work to be completed by the CONTRACTOR shall be considered incidental to the project and no additional compensation will be given to the CONTRACTOR for the completion of this work.

3.12 UTILITIES

The OWNER is responsible for determining the location of all utilities and for protecting them from damage during construction. Should any marked utility be damaged or disturbed, the CONTRACTOR shall immediately notify the OWNER and ENGINEER of the damaged utility. All damage that results from work under the contract shall be promptly repaired at the expense of the CONTRACTOR. The source of electrical power to operate CONTRACTOR's equipment shall be the CONTRACTOR's responsibility. The CONTRACTOR will also be responsible for obtaining water for drilling as needed.

3.13 WELL GROUTING

Well grouting shall consist of the following:

- The cased well must be grouted with at least 10 feet of continuous grouting, within the zone from 10 to 20 feet below the ground surface. Prior to grouting, the annular space should be flushed to assure that the space is open and ready to receive the sealing material. Grouting shall be done in one continuous operation in which the annular space is filled, and any exterior casing removed. Grout placement shall be by the positive placement method as outlined in Section 48.040-000-000 of the EPA's Manual of Water Well Construction Practices.
- Acceptable Grouting Materials:
 - bentonite grout slurry or granules
 - Neat Cement Grout.
 - A mixture of Portland cement (ASTM C150) and not more than seven (7) gallons of clean water per bag (one cubic foot or 94 pounds) of cement, shall be used.
 - The use of special cements, bentonite to reduce shrinkage, or other admixtures (ASTM C494) to reduce permeability, increase fluidity, and/or control time of set, and the composition of the resultant slurry must be approved by the ENGINEER.

3.14 WELL HEAD COMPLETION

Well head completions shall consist of the following:

The well casing (stick-up) will protrude a minimum of 24 inches above the existing grade and be completed with a locking vandal-proof sanitary seal (well cap). The CONTRACTOR shall submit a proposed design of the cap prior to installation.

All wells should be labeled with a durable form of construction with information upon completion.

The construction information source should be secured to the well casing and contain the following information:

- ✓ The name of the well driller, pump installer, and business affiliations;
- ✓ The date the well was completed;
- ✓ The total well depth;
- ✓ The total depth of casing;
- ✓ The location and type of well_completion;
- ✓ Static water level_(SWL) below the top of the casing;
- ✓ Well yield; and
- ✓ Height of casing_above finished grade.

The CONTRACTOR shall provide grading of the ground surface for at least 10 feet radially from the well to provide drainage away from the well.

3.15 WELL ABANDONMENT

As determined by the ENGINEER and OWNER, the well may be abandoned if there is insufficient capacity, unsatisfactory water quality, poor alignment, loss of tools or other unforeseen problems. The methods of well abandonment shall meet the requirements and procedures in accordance with the procedures outlined in the publication titled "Alaska Best Management Practices for Maintaining and Decommissioning Water Wells and Boreholes" referenced above in Section 1.2. A well abandoned due to contamination, poor alignment, loss of tools or any other cause attributed to the CONTRACTOR shall be at the CONTRACTOR's expense.

3.16 MOBILIZATION, DEMOBILIZATION, AND CLEANUP OF SITE

To facilitate coordination of work, the CONTRACTOR shall notify the OWNER's Representative at least 48 hours in advance of mobilizing and demobilization. The CONTRACTOR shall always perform his work in a manner to ensure minimal obstruction to traffic and minimal inconvenience to the general public and the OWNER.

It is anticipated the well site will be located approximately 50-feet from the airport haul road. The CONTRACTOR shall provide "tundra mats" sufficient to support the drill rig to the proposed location for the well. It is estimated that a minimum of 60-feet long "tundra mats" will be needed to access the well drill site.

During the progress of work, the CONTRACTOR shall protect all existing vegetation, structures, roads, ditches, culverts, signposts, fences, driveways, and similar items. If the CONTRACTOR must remove or disturb these facilities, provision shall be made to maintain a temporary facility serving the same purpose as that which it was intended to replace. These provisions shall include:

- Cuttings, drillings or other debris shall not be washed into drainage structures that might cause clogging of drainage in any way.
- The CONTRACTOR shall restore all areas disturbed by construction to a condition at least equal to that existing prior to construction. Excess construction materials, equipment, tools, waste excavation, and rubbish shall be removed. Excavated areas shall be graded to provide drainage as required by the drawings and specifications, or in the absence of specific requirements, to restore original drainage patterns in existence prior to construction.

3.17 MEASUREMENT AND PAYMENT

The contract price for each item shall constitute full compensation for furnishing all labor, equipment, appliances, and material, and performing all operations necessary to construct and complete the items in accordance with the requirements of the specifications as full compensation notwithstanding that minor features of work to complete the item may not be mentioned.

Deviation in the actual quantities either above or below the estimated quantities shown for each item shall not be a basis for a claim for adjustment in any of the contract unit prices. Work paid for under one item will not be paid for under another item.

ITB # VSW-TNK-2019-23 6-Inch Diameter Water Supply Well, Tununak, AK Mobilization and demobilization: All required. Payment for mobilization for well construction will be made at the lump sum prices named in the Bid Schedule under Item No. 1, which price shall constitute full compensation for completion of all transportation of CONTRACTOR's equipment, personnel, materials, tools, supplies, and the provision of all facilities for temporary power, water, fire protection, communications, and storage at the project site. This item also includes obtaining all necessary permits required by local, state, and federal governmental agencies. Bids of mobilization greater than 50 percent of the contract amount will be deemed unbalanced and may be cause for rejection. If, however, the CONTRACTOR can prove to the ENGINEER by quotes or other written documentation that the cost of mobilization with its share of overhead and profit is more than 50 percent of the contract amount, the ENGINEER will allow payment for mobilization commensurate with documented cost with overhead and profit. For purposes of pay, 80 percent of the lump sum price shall be paid for mobilization with the remaining 20 percent paid for demobilization, in accordance with the Contract Documents. This item includes protection of the well head and cleanup of each site as specified. It is anticipated that four wells will be installed. The number and location within the site area may vary. Payment for this item shall be on a lump sum basis.

Driven Casing: The number of linear feet shall be determined by the depth of the hole drilled and accepted, measured from the existing surface to the bottom of that size casing at its deepest point prior to pull-back for screen installation. Any open hole drilled beyond the bottom of the casing shall be measured for payment purposes, at 75% of the distance of open hole drilled. No consideration will be given for oversize casing unless called out by the bid schedule. The CONTRACTOR must be prepared to drill through, screen, and test pump any aquifers located within the depth called for by the bid schedule. It is anticipated that the well will be 100 to 200-feet in depth. The CONTRACTOR will be paid for a minimum depth of 100-feet even if the well's depth is less than 100-feet; however, the maximum well depth is not expected to exceed 200-feet in depth unless authorized by the OWNER and ENGINEER. Payment for this item shall be on a linear foot basis.

Drill and Case 6-Inch (Nominal) Diameter Water Wells – Boring For Screened Casing: The number of linear feet shall be determined by the depth of the hole drilled and accepted. Measurement of the lineal foot of depth drilled shall begin where excavation begins in the borehole pertaining to this task. No consideration will be given for oversize casing unless called out by the bid schedule. The CONTRACTOR must be prepared to drill through, screen and test pump any aquifers located within the depth called for by the bid schedule. Payment for this item shall be on a linear foot basis.

Furnish and Install 6-Inch (Nominal) Diameter Well Screen: The number of linear feet of well screen shall be determined by the length of well screen measured. For bid solicitation purposes, a length of 10-feet for the well screen will be used. All backfill, and gravel pack material used, and its placement shall be considered incidental to the cost of this pay item. Payment for this item shall be on a linear foot basis.

Well Development: This item shall be based on the actual number of hours required to fully develop each well as specified, including time for removal of sand. Additional testing required by the

ENGINEER which utilizes the CONTRACTOR's drilling rig and personnel shall also be paid for under this item. Payment for this item shall be on a per hour basis.

Test Pumping: Covers the actual number of hours of acceptable test pumping performed as specified. This item shall include installation and removal of the test pump including: test pump installation, hookup and removal temporary pipeline required for test pump, water, and disposal. Disinfection of the well shall also be included in this item. Payment for this item shall be on a per hour basis.

Install Well Grouting, Well Casing Stick-Up, and Locking Cap: This item includes all grouting as specified, well casing above grade (stick-up), and specified locking cap. Erosion control measures shall be included in this item. Payment for this item shall be on a lump sum basis.

Well Abandonment: This item includes all labor, materials, and equipment required to properly abandon a well in accordance with the Specifications provided herein. This bid item shall be paid on a lump sum basis for each well abandoned due to no cause of the CONTRACTOR.

Added Alternate: All required subject to availability of funding to furnish (provide and deliver) 200 lineal feet of 6-inch diameter well casing. Payment for mobilization will be made at the lump sum price named as an "Added Alternate" in the Bid Schedule, which price shall constitute full compensation for completion of all transportation of materials and storage at the project site.

SUPPLEMENTAL CONTRACT REQUIREMENTS

1. BID SCHEDULE

1.1. This is a unit rate contract. Refer to the Bid Schedule for the schedule of values.

2. CONSTRUCTION MANAGEMENT

2.1. The work shall be laid out by the Engineer and performed by the Contractor. Daily pay items shall be submitted by the Contractor daily on a form supplied by the Engineer. The form shall be reviewed by the Engineer on the next working day, and if approved, signed by the Engineer and Contractor. Items not identified will not be paid.

3. MBE/WBE Preference

Minority Business Enterprise (MBE) and Women's Business Enterprise (WBE).

This procurement is funded in part or fully through federal grants or cooperative agreements. It is a national policy to award a fair share of contracts to Minority Firms and Women's Business Enterprises through affirmative action. The negotiated Federal "Fair Share" percentage for **fiscal years 2018 through 2019** is 3.67% MBE and 1.54% WBE. This solicitation incorporates a five point preference for all qualified minority firms and women's business enterprises.

In order to be deemed a bona fide Minority Business Enterprise (MBE) or Women's Business Enterprise (WBE) a firm must be an independent business concern which is a least fifty-one percent (51%) owned and controlled by minority group members or women.

It is the responsibility of the offeror to include in the bid their qualifications and/or of the qualifications of their subcontractors for this preference. It is also the responsibility of the offeror claiming eligibility for this preference to pledge in the proposal that the eligible subcontractor will be **guaranteed** at least 5.21% of the proposed work.

Following is an example of how the preference points will be calculated for qualifying businesses:

MBE/WBE Offeror's Preference

[STEP 1]

Determine the number of points available to MBE/WBE eligible offerors under this preference.

Total number of points available in this example situation = 100 Points

100 x 5% = 5

Total Points MBE/WBE Offeror's Number of Points Available Percentage Preference to Eligible Offerors

Under MBE/WBE Preference

[STEP 2]

Add the preference points to the qualified MBE/WBE offers. In a hypothetical situation, there are three (3) offerors. After being evaluated, each received the following points:

Offeror #1 95 Points Offeror #2 90 Points Offeror #3 92 Points

Before preference points are calculated, offeror #1 is the apparent winner. However, in this hypothetical situation, offeror #2 and offeror #3 are eligible for the MBE/WBE preference. After adding five (5) points to their scores, offeror #3 is the new apparent winner, with ninety-seven (97) points.

4. USDA-RD REQUIREMENTS

4.1. Rural Alaska Village Grant (RAVG) Agreement

4.1.1.Each recipient community must execute a Rural Alaska Village program Grant Agreement. The Agreement must be signed by the authorized official in each community. The signature of the official must be attested. The articles of incorporation must support the title of the authorized and attesting official.

4.2. Contracts and Procurement

4.2.1.Contracting and procurement activities must be conducted using the Department policies, procedures, and methods.

4.3. Civil Rights Act

4.3.1.The Contractor shall comply with Civil Rights and Equal Opportunity statutes including but not limited to: Title VI of the Civil Rights Act of 1964 (42 U.S. C. 2000d et seq.) as outlined in 7 CFR 1901 subpart E of this title; Section 504 of the Rehabilitation Act of 1973; The Americans with Disabilities Act (ADA) of 1990; and Age Discrimination Act of 1975.

4.4. Property

4.4.1.The Contractor shall comply with the requirements set forth in 7 CFR Part 21 of the Uniform Relocation and Real Property Acquisition Policies Act (42 U.S.C. 4601 et seq.)

4.5. Floodplains

4.5.1.The Contractor shall follow the eight-step decision making process referenced in Section 2 (a) of Executive Order (E.O) 11988, Floodplain Management, when undertaking actions located in floodplains. Pursuant to E.O. 11988, the IHS uses a Class Review process to exclude certain actions from further review under the eight- step process. For all actions that do not qualify for IHS Class Review, the eight- step process shall be completed. All practicable measures to minimize harm to floodplains and people shall be

- followed, including elevating a new water or wastewater facility at least 1 foot above the recommended building height as determined by the Army Corp of Engineers.
- 4.5.2.If an area has been designated by the Federal Emergency Management Administration, flood insurance shall be required for facilities located in floodplains. If a community is not participating in the National Flood Insurance Program the requirement does not apply.

4.6. Government Debarment and Suspension and Drug Free Workplace

4.6.1.The Contractor shall comply with the requirements set forth in the U.S. Department of Agriculture regulations 7 CFR part 3017 and RD Instruction 1940 M.

4.7. Intergovernmental Review

4.7.1. The Contractor shall comply with the requirements set forth in Executive Order 12372 (3 CFR, 1983 Comp., p 197), which requires intergovernmental consultation with State and local officials. The requirements are set forth in U. S. Department of Agriculture regulations 7 CFR Part 3015, Subpart V, and RD Instruction 1940 J.

4.8. Environmental Review

4.8.1. Any planning or design project where a geotechnical investigation will be completed must receive the appropriate level of environmental review.

END OF SUPPLEMENTAL CONTRACT REQUIREMENTS

Section 7

Attachments List

- A. <u>Bid Schedule and Contractor's Questionnaire</u>— (5 pages)
- B. <u>Drawings Tununak Bay Location Map (1 Page)</u>
- C. <u>Drawings Well Drilling Locations (1 Page)</u>

Appendices

- A. Standard General Conditions of the Construction Contract (39 Pages)
- B. Federal Debarment Certification Form (2 Pages)
- C. Certification and Disclosure Regarding Payments to Influence Certain Federal Transactions (3 Pages)
- D. Subcontractor List (2 Pages)

Attachment A

1. BID SCHEDULE

6-INCH DIAMETER WATER SUPPLY WELL, TUNUNAK, AK

ITEM #	PAY ITEM NAME AND UNIT BID PRICE IN WORDS	ESTIMATED QUANTITY	UNIT BID PRICE	TOTAL BID PRICE
1	Mobilization, demobilization, site cleanup and provide "tundra mats" as specified.	All Required	\$	\$
	(Lump Sum) in words		Lump Sum	
2	Drill and install well casing for 6-inch (nominal) diameter water supply well – boring for blank well casing/driven casing. (Estimated quantity based on a well having a depth ranging from 100 to 200-feet described in Specifications).	200	\$	\$
	(Per Linear Foot) in words		Per Linear Foot	
3	Furnish and install 6-inch (nominal) diameter well screen including filter pack or formation stabilizer and bail bottom or bottom plate fitting and a riser. (Estimated quantity based on a well with 10-foot screen section).	10	\$	\$
	(Per Linear Foot) in words		Per Linear Foot	
4	Well development. (Estimated quantity based on 12 hours of well development).	12	\$	\$
	(Per Hour) in words		Per Hour	
5	Test pumping. (Estimated quantity based on 24 hours of well pumping). Includes well recovery measurements and all costs associated with installation and removal of test pumps.	24	\$	\$
	(Per Hour) in words		Per Hour	
6	Disinfect completed well and install well grouting, well casing stickup, and locking sanitary seal (well cap).	All Required	\$	\$
	(Lump Sum) in words	•	Lump Sum	
7	Well Abandonment, if required.	All Required	\$	\$
	(Lump Sum) in words	7111 Required	Lump Sum	
TOTAL	BASE BID		\$	
(Total B	id Price In Words)			
	O ALTERNATE: Provide & Deliver 200 LF 6-inch Diamete	er Well Casing	\$	
(Lump S	um) in words			

The quantities shown in the Bid, whether for a unit price contract, or a combination of a lump sum contract and unit price contract, are approximate only and should not be taken as either representations or warranties. Since quantities in the Contract Documents are estimates only, actual quantities may increase or decrease without constituting a change in the Work.

The OWNER reserves the right to award to the low responsive and responsible Bidder based on the total bid amount if that amount is within available funding; however, if that amount is greater than available funding, the OWNER reserves the right to the low responsive and responsible Bidder based on the Base Bid without the alternate as funding allows.

ward to the

Name of Bidding Firm: _		
Signature:		
Name and Title:		
Date:		

2. CONTRACTOR'S QUESTIONNAIRE

Project Name and Location: 6-Inch Diameter Water Supply Well, Tununak, Alaska

FINANCIAL

- 1. Have you ever failed to complete a contract on account of insufficient resources? If so, explain:
- 2. Have you made arrangements to finance the work? If so, with whom and for what amount?

EQUIPMENT

1. List in the table below the equipment which you have available for the work that you propose to do. This equipment should be listed in detail. General statements will not be accepted. Attach additional pages if necessary.

NO.	ITEMS	ТҮРЕ	SIZE/ CAPACTITY	PRESENT VALUE

2.	Do you understand that, if you are awarded this contract, you may be required to use any or all the equipment listed on the work covered by this contract?				
3.	Do you propose to purchase any equipment for use on this project should the contract be awarded to you?				
	11 so, state type, quantity, and approximate cost.				
4.	Do you propose to rent any equipment for this work? If so, state type, quantity, and reason for renting.				
5.	Provide the type and model of drill rig(s) proposed for use on this project.				
6.	Provide the type, model, and capacity of test pump proposed for use on this project.				
7.	What method of well development do you intend to use on this project and provide a				
	description of the equipment that will be used for well development.				
8.	Do you intend to subcontract any portion of the work contained in this project? If so, list the types and portions of the work to be subcontracted.				
	Approximate total value \$				
	Percent of total bid				

EXPERIENCE

- 1. Have you performed well drilling contracts or subcontracts involving the installation of drinking water wells for the State of Alaska Village Safe Water Program or the Village of Tununak? If so, briefly describe project on separate page with dates, scope of work, etc.
- 2. List five well drilling projects in the following table that were completed in the past five years involving work of similar type and complexity. Include the dates of completion, scope of work, and total contract amount for each project.

Project	Well Diameter (inches)	Total Well Depth (feet)	Well Capacity (gpm)	Reference Information
,			(SI)	
I hereby certify that the al	oove stateme	nts are true and	l complete.	
Signature				
Name of Contractor				
Title of Person Signing				

Date



Figure 1
Vicinity and Location (Inset)
Source: ADCRA Mapping

0.000.000.000.000





Figure 2 Tununak, Alaska Proposed Well Drilling Locations

APPENDIX A

STANDARD GENERAL CONDITIONS OF THE

CONSTRUCTION CONTRACT

1.1 BASIC DEFINITIONS

1.1.1 The Contract Documents. The contract documents consist of the construction services standard contract between the Owner and Contractor (hereinafter the agreement), conditions of the contract (general conditions, supplementary and other conditions), invitation to bid, addenda issued prior to execution of the contract, drawings, specifications, and other documents listed in the agreement and modifications issued after execution of the contract. A modification is (1) a written amendment to the contract signed by both parties, (2) a change order, (3) a construction change directive, and (4) a written order for a minor change in the work issued by the Engineer.

In the event of a conflict of provisions, the following order of precedence shall apply in resolving which provisions control:

- 1. Any written amendment to the construction services executed standard contract, with the more recent amendment taking precedence over the less recent amendment;
- 2. Construction services standard contract;
- 3. Standard General Conditions of the Construction Contract;
- 4. Solicitation document including all attachments and addendums;
- 5. Contractors bid including all attachments.
- 1.1.2 The Contract. The contract documents form the contract for construction. The contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. Except as set forth in paragraph 7.3 and 7.4, the contract may be amended or modified only by a change order. The contract documents shall not be construed to create a contractual relationship of any kind (1) between the Engineer and Contractor, (2) between the Owner Representative and a subcontractor of any tier, or (3) between any persons or entities other than the Owner Representative and Contractor.
- 1.1.3 The Work. The term "work" means the construction and services required by the contract documents, whether completed or partially completed, and includes the result of performing or providing, all necessary equipment, materials, supplies, tools, freight, labor, supervision, water, heat, utilities, and transportation provided or to be provided by the Contractor to fulfill the Contractor's obligations. The work may constitute the whole, or a part of the project.
- 1.1.4 <u>The Project</u>. The project is the total construction of which the work performed under the contract documents may be the whole or a part and which may include construction by the Owner Representative or by separate Contractors.
- 1.1.5 The Drawings. The drawings are the graphic and pictorial portions of the contract documents,

wherever located and whenever issued, showing the design, location and dimensions of the work, generally including plans, elevations, sections, details, schedules and diagrams.

- 1.1.6 <u>The Specifications</u>. The specifications are that portion of the contract documents consisting of the written requirements for materials, equipment, construction systems, standards and workmanship for the work, and performance related services.
- 1.1.7 <u>Approved</u>. When the words "approved," "satisfactory," "proper," or "as directed" are used, approval by the Engineer shall be understood.
- 1.1.8 <u>Provide</u>. When the word "provide" is used, it shall mean to properly fabricate, complete, transport, deliver, install, erect, construct, test and furnish all necessary equipment, materials, supplies, tools, freight, labor, supervision, water, heat, utilities, transportation, and all other items necessary to properly complete and place, ready for operation or use under the terms of the specifications.
- 1.1.9 <u>Knowledge</u>. The terms "knowledge," "recognize," and "discover," when used in the contract documents in reference to the Contractor, shall mean that the Contractor knows (or should know), recognizes (or should recognize), and discovers (or should discover) in exercising the care, skill and diligence required by the contract documents.
- 1.1.10 <u>Reasonably Inferable</u>. The term "reasonably inferable" shall mean reasonably inferable by a Contractor familiar with the project and exercising the care, skill, and diligence required of the Contractor by the contract documents.
- 1.1.11 <u>Persistently Fails</u>. The phrase "persistently fails," as used in reference to the Contractor, shall be interpreted to mean any combination of acts and omissions that cause the Owner Representative or the Engineer to reasonably conclude that the Contractor will not complete the work within the contract term, for the contract sum or in substantial compliance with the requirements of the contract documents.
- 1.1.12 <u>The Owner</u>. The "Owner" is defined as the City or the Community.
- 1.1.13 <u>The Owner's Representative.</u> The "Owner's Representative" is defined as the Village Safe Water Program.
- 1.1.14 <u>The Engineer.</u> The "Engineer" is defined as the Village Safe Water Program or its authorized representative.
- 1.1.15 The term "Contractor" means the Contractor or the Contractor's authorized representative.
- 1.1.16 A "Subcontractor" is a person or entity who has a direct contract with the Contractor to perform a portion of the work at the site.

1.2 EXECUTION, CORRELATION AND INTENT

1.2.1 The intent of the contract documents are to include all items necessary for the proper execution

and completion of the work by the Contractor. The contract documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the contract documents and reasonably inferable from them as being necessary to produce the intended results.

- 1.2.2 Organization of the specifications into divisions, sections and articles, and arrangement of drawings shall not control the Contractor in dividing the work among subcontractors or in establishing the extent of work to be performed by any trade.
- 1.2.3 Unless otherwise stated in the contract documents, words which have well known technical or construction industry meanings are used in the contract documents in accordance with such recognized meanings.
- 1.2.4 In the event of inconsistencies within or between parts of the contract documents, or between the contract documents and applicable standards, codes and ordinances, the Contractor shall (1) provide the better quality or greater quantity of work, or (2) comply with the more stringent requirement in accordance with the Engineer's interpretation. The terms and conditions of this paragraph, however, shall not relieve the Contractor of any of the obligations set forth in paragraphs 3.2 and 3.7.
- 1.2.5 On the drawings, given dimensions shall take precedence over scaled measurements.
- 1.2.6 When requested by the Engineer or specified, support test data shall be submitted to substantiate compliance of any specified product with the particular standard or specification indicated in the contract documents.

1.3 INTERPRETATION

1.3.1 In the interest of brevity the contract documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

ARTICLE 2

OWNER REPRESENTATIVE

2.1 INFORMATION AND SERVICES REQUIRED OF THE OWNER REPRESENTATIVE

- 2.1.1 The Owner's Representative shall secure and pay for any necessary state and federal environmental permits required for construction.
- 2.1.2 The Owner Representative's action or non-action shall not relive the Contractor of the responsibility to comply with the specifications.
- 2.1.3 Information or services under the Owner Representative's control shall be furnished by the Owner Representative with reasonable promptness to avoid delay in orderly progress of the work.

2.2 OWNER REPRESENTATIVE'S RIGHT TO STOP THE WORK

2.2.1 If the Contractor fails to correct work which is not in accordance with the requirements of the contract documents as required by paragraph 12.2 or persistently fails to carry out work in accordance with the contract documents, the Owner Representative, by written order signed personally or by an agent specifically so empowered by the Owner Representative in writing, may order the Contractor to stop the work, or any portion thereof, until the cause for such order has been eliminated. However, the right of the Owner Representative to stop the work shall not give rise to a duty on the part of the Owner Representative to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by subparagraph 6.1.3.

2.3 OWNER REPRESENTATIVE'S RIGHT TO CARRY OUT THE WORK

2.3.1 If the Contractor defaults or neglects to carry out the work in accordance with the contract documents and fails within a seven day period after receipt of written notice from the Owner Representative to commence and continue correction of such default, or neglect with diligence and promptness, the Owner Representative may, without prejudice to other remedies correct such deficiencies. In such case an appropriate change order shall be issued deducting from payments then or thereafter due the Contractor the cost of correcting such deficiencies, including compensation for the Engineer's additional services and expenses made necessary by such default, neglect or failure. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner Representative.

2.4 NO CONTROL BY OWNER REPRESENTATIVE

2.4.1 In no event shall the Owner Representative have control over, charge of, or any responsibility for construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the work, notwithstanding any of the rights and authority granted the Owner Representative in the contract documents.

ARTICLE 3

CONTRACTOR

3.1 **DEFINITION**

- 3.1.1 The Contractor shall perform the work in accordance with the contract documents and submittals approved pursuant to paragraph 3.11.
- 3.1.2 The Contractor shall not be relieved of obligations to perform the work in accordance with the contract documents either by activities or duties of the Engineer in the Engineer's administration of the contract, or by tests, inspections, or approvals required or performed by persons other than the Contractor.

3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

- 3.2.1 Since the contract documents are complementary, before starting each portion of the work, the Contractor shall carefully study and compare the various drawings and other contract documents relative to that portion of the work as well as the information furnished by the Owner Representative pursuant to subparagraph 2.2. These obligations are for the purpose of facilitating construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the contract documents. However, any errors, inconsistencies, or omissions discovered by the Contractor shall be reported promptly to the Engineer as a request for information in such form as the Engineer may require.
- 3.2.2 If the Contractor believes that additional cost or time is involved because of clarifications or instructions issued by the Engineer in response to the Contractor's notices, or requests for information pursuant to subparagraphs 3.2.1 and 3.2.3, the Contractor shall make claims as provided in subparagraphs 4.3.6 through 4.3.8. If the Contractor fails to perform the obligations of subparagraphs 3.2.1, the Contractor shall pay such costs and damages to the Owner Representative as would have been avoided if the Contractor had performed such obligations. The Contractor shall not be liable to the Owner Representative or Engineer for damages resulting from errors, inconsistencies, or omissions in the contract documents or for differences between field measurements or conditions and the contract documents unless the Contractor recognized such error, inconsistency, omission, or difference and knowingly failed to report it to the Engineer.
- 3.2.3 The Contractor shall take field measurements and verify field conditions and shall carefully compare such field measurements and conditions and other information known to the Contractor with the contract documents before commencing activities. Errors, inconsistencies or omissions discovered shall be reported to the Engineer immediately.
- 3.2.4 Except as to any reported errors, inconsistencies or omissions, and as to concealed or unknown conditions as defined in paragraph 4.3.5, by executing the agreement, the Contractor represents the following:
 - 3.2.4.1 The contract documents are sufficiently complete and detailed for the Contractor to (1)

perform the work required to produce the results intended by the contract documents, and (2) comply with all the requirements of the contract documents.

3.2.4.2 The work required by the contract documents including without limitation, all construction details, means, methods, procedures and techniques, use of materials, supplies, selection of equipment, and requirements of product manufacturers are consistent with: (1) good and sound practices within the construction industry; (2) generally prevailing and accepted industry standards applicable to the work; (3) requirements of any warranties applicable to the work; and (4) all laws, ordinances, regulations and/or rules which bear upon the Contractor's performance of the work.

3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- 3.3.1 The Contractor shall supervise and direct the work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over construction means, methods, techniques, sequences and procedures and for coordinating all portions of the work under the contract, unless contract documents give other specific instructions concerning these matters. If the contract documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the job site safety thereof and, except as stated below, shall be fully and solely responsible for the job site safety of such means, methods, techniques, sequences, or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner Representative and Engineer and shall not proceed with that portion of the work without further written instructions from the Engineer.
- 3.3.2 The Contractor shall be responsible to the Owner Representative for acts and omissions of the Contractor's employees, subcontractors of whatever tier and their agents and employees, and other persons performing portions of the work.
- 3.3.3 The Contractor shall be responsible for inspection of portions of work already performed under the contract to determine that such portions are in proper condition to receive subsequent work.

3.4 LABOR AND MATERIALS

- 3.4.1 Unless otherwise provided in the contract documents, the Contractor shall provide and pay for all necessary equipment, materials, supplies, tools, freight, labor, supervision, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the work, whether temporary or permanent and whether or not incorporated or to be incorporated in the work.
- 3.4.2 The Contractor may make substitutions only with the consent of the Owner Representative, after evaluation by the Engineer and in accordance with a change order.
- 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the work. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them.

- 3.4.4 <u>Prevailing Wages</u>. The Contractor must comply with all requirements of Alaska Statute 36.05, entitled Public Contracts, Wage & Hour Administration, including the latest State of Alaska Department of Labor & Workforce Development Laborers and Mechanics Minimum Rates of Pay Pamphlet No. 600. http://www.labor.state.ak.us/lss/pamp600.htm is the website link for the current document. The Contractor is responsible for ensuring they use the most up-to- date version of Pamphlet No. 600.
- 3.4.5 <u>Davis Bacon Act.</u> The Contractor must comply with all requirements of the Davis Bacon Act (40 U.S.C. 3141-3144, 3146-3148 and 40 U.S.C. 276a to 276a-7).
- 3.4.6 <u>Contract Work Hours and Safety Standards Act.</u> The Contractor must comply with all requirements of the Contract Work Hours and Safety Standards Act (40 U.S.C. 3701-3708), as supplemented by the Department of Labor regulations (29 CFR part 5).
- 3.4.7 <u>Byrd Anti-Lobbying Amendment.</u> The Contractor must comply with all requirements of the Byrd Anti-Lobbying Amendment (31 U.S.C 1352) (Appendix C). Appendix C must be completed and submitted prior to contract award.
- 3.4.8 <u>Copeland Anti-Kick Back Act.</u> The Contractor must comply with all requirements of the Copeland Anti-Kick Back Act (18 U.S.C. 874 and 40 U.S.C 3145) as supplemented in Department of Labor regulations (29 CFR, Part 3).
- 3.4.9 <u>Solid Waste Disposal Act.</u> The Contractor must comply with all requirements of section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act.
- 3.4.10 <u>Clean Air Act and Water Pollution Control Act.</u> The Contractor must comply with all requirements of the Clean Air Act (42 U.S.C. 7401-7671q) including section 114 and section 308 of the Water Pollution Control Act (33 U.S.C. 1251-1388).
- 3.4.11 <u>Clean Water Act.</u> The Contractor must comply with all requirements of section 215 (Requirements for American Materials) of the Clean Water Act (33 U.S.C. 1251 et seq.) and implementing EPA regulations. The Contractor agrees that preference will be given to domestic construction materials by the Contractor, subcontractors, materialmen, and suppliers.
- 3.4.12 <u>Equal Employment Opportunity.</u> The Contractor must comply with all requirements of Executive Order 11246 (3 CFR, 1966 Comp., p. 339), entitled, "Equal Employment Opportunity," as amended by Executive Order 11375 (3 CFR, 1968 Comp., p. 321), and as supplemented by the Department of Labor regulations 41 CFR chapter 60.
- 3.4.13 <u>Drug Free Workplace Act of 1988.</u> The Contractor must comply with all requirements of the regulations implementing Sections 5151-5160 of the Drug Free Workplace Act of 1988 (Pub. L. 100-690, Title V, Subtitle D; 41 U.S.C, 701 et seq.), 7 CFR Part 3017, Subpart F, Section 3017.600, Purpose. The regulations were published as Part II of the January 31, 1989 Federal Register (pages 4947-4952).
- 3.4.14 <u>Americans with Disabilities Act (ADA) of 1990.</u> The Contractor must comply with all requirements of the Americans with Disabilities Act of 1990 (42 U.S.C 12101 et seq.) and the regulations issued thereunder by the federal government. Services or activities furnished to the general public on behalf of the state must be fully accessible. This is intended to ensure that agencies are in accordance

with 28 CFR Part 35 Section 35.130 and that services, programs or activities furnished to the public through a contract do not subject qualified individuals with a disability to discrimination based on the disability.

- 3.4.15 <u>Civil Rights Act of 1964.</u> The Contractor must comply with all requirements of Title VI of the Civil Rights Act of 1964 (42 U.S.C. 2000d et seq.) as outlined in 7 CFR 1901 subpart E.
- 3.4.16 <u>Rehabilitation Act of 1973.</u> The Contractor must comply with all requirements of section 504 of the Rehabilitation Act of 1973, as amended (29 U.S.C. 794).
- 3.4.17 <u>Age Discrimination Act of 1975.</u> The Contractor must comply with all requirements of the Age Discrimination Act of 1975 (42 U.S.C. 6101 et seq.).
- 3.4.18 <u>Federal Debarment, Suspension, Ineligibility and Voluntary Exclusion</u>. Expenditures from this contract may involve federal funds. The U.S. Department of Labor requires all state agencies that are expending federal funds to have a certification filed in the bid (by the bidder) that they have not been debarred or suspended from doing business with the federal government. Certification regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion Lower Tier Covered Transactions (Appendix B) **must be completed and submitted prior to award.** This certification is required by the regulations implementing Executive Order 12549, Debarment and Suspension, 29 CFR Part 98, Section 98.510, Participant's responsibilities. The regulations were published as Part VII of the May 26, 1988 Federal Register (pages 19160-19211).
- 3.4.19 The Contractor must comply with all applicable local, Federal and State statutes, regulations, ordinances and codes, whether or not specifically mentioned herein.
- 3.4.20 <u>Human Trafficking</u>. By signature on their bid, the bidder certifies that the bidder is not established and headquartered or incorporated and headquartered in a country recognized as Tier 3 in the most recent United States Department of State's Trafficking in Persons Report.

The most recent United States Department of State's Trafficking in Persons Report can be found at the following website: http://www.state.gov/g/tip/

The Contractor must comply with this requirement, failure to comply with this requirement will cause the state to reject the bid as non-responsive, or cancel the contract.

3.5 WARRANTY

3.5.1 The Contractor warrants to the Owner Representative and Engineer that materials and equipment furnished under the contract will be of good quality and new unless otherwise required or permitted by the contract documents, that the work will be free from defects not inherent in the quality required or permitted, and that the work will conform with the requirements of the contract documents. Work not conforming to these requirements, including substitutions not properly approved and authorized, shall be considered defective. Work that does not conform to applicable laws, ordinances, or building codes shall be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, modifications not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear under normal usage. If required by the

Engineer, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

3.5.2 The Contractor shall perform the work in such manner so as to preserve any and all manufacturer's warranties. The Contractor shall assign to the Owner Representative at the time of final completion of the work any and all manufacturer's warranties relating to materials and labor used in the work.

3.6 PAYMENT OF TAXES

3.6.1 The Contractor shall pay sales, consumer, use and similar taxes for the work or portions thereof provided by the Contractor which are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

3.7 PERMITS, FEES AND NOTICES

3.7.1 Except as otherwise provided elsewhere in the contract documents, the Contractor shall pay for any permits and governmental fees, licenses and inspections necessary for proper execution and completion of the work.

3.8 SUPERINTENDENT

3.8.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the project site during performance of the work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor.

3.9 CONTRACTOR'S CONSTRUCTION SCHEDULES

- 3.9.1 The Contractor, promptly after being awarded the contract, shall prepare and submit for the Owner Representative's and Engineer's information a Contractor's construction schedule for the work. The schedule shall not exceed time limits current under the contract documents, shall be revised at appropriate intervals as required by the conditions of the work and project, shall be related to the entire project, and shall provide for expeditious and practicable execution of the work.
- 3.9.2 The Contractor shall prepare and keep current, for the Engineer's approval, a schedule of submittals which is coordinated with the Contractor's construction schedule and allows the Engineer a reasonable time to review submittals.
- 3.9.3 The Contractor shall conform to the most recent schedules.

3.10 DOCUMENTS AND SAMPLES AT THE SITE

3.10.1 The Contractor shall maintain at the site for the Owner Representative one record copy of the drawings, specifications, addenda, change orders and other modifications, in good order and marked currently to record changes and selections made during construction, and in addition approved shop drawings, product data, samples and similar required submittals. These shall be available to the Engineer and shall be delivered to the Engineer for submittal to the Owner Representative upon completion of the work.

3.11 SHOP DRAWINGS, PRODUCT DATA, AND SAMPLES

- 3.11.1 Shop drawings are drawings, diagrams, schedules, and other data specially prepared for the work by the Contractor or a subcontractor of whatever tier, manufacturer, supplier, or distributor to illustrate some portion of the work.
- 3.11.2 Product data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams, and other information furnished by the Contractor to illustrate materials or equipment for some portion of the work.
- 3.11.3 Samples are physical examples which illustrate materials, equipment, or workmanship and establish standards by which the work will be judged.
- 3.11.4 Shop drawings, product data, samples, and similar submittals are not contract documents. The purpose of their submittal is to demonstrate for those portions of the work for which submittals are required the way the Contractor proposes to conform to the information given and the design concept expressed in the contract documents. Review by the Engineer is subject to the limitations of subparagraph 4.2.7. Informational submittals upon which the Engineer is not expected to take responsive action may be so identified in the contract documents. Submittals which are not required by the contract documents may be returned by the Engineer without action.
- 3.11.5 The Contractor shall review, approve, and submit to the Engineer shop drawings, product data, samples and similar submittals required by the contract documents with reasonable promptness and in such sequence as to cause no delay in the work or in the activities of the Owner Representative or of separate Contractors. Submittals made by the Contractor which are not required by the contract documents may be returned without action.
- 3.11.6 By approving and submitting shop drawings, product data, samples, and similar submittals, the Contractor represents that the Contractor has determined and verified materials, field measurements, and field construction criteria related thereto, or will do so, and has checked and coordinated the information contained within such submittals with the requirements of the work and of the contract documents.
- 3.11.7 The Contractor shall perform no portion of the work requiring submittal and review of shop drawings, product data, samples, or similar submittals until the respective submittal has been approved by the Engineer. Such work shall be in accordance with approved submittals.
- 3.11.8 The work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the contract documents by the Engineer's approval of shop drawings, product data, samples, or similar submittals unless the Contractor has specifically informed the Engineer in writing of such deviation at the time of submittal and the Engineer has (1) given written approval to the specific deviation as a minor change in the work, or (2) a change order or construction change directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in shop drawings, product data, samples, or similar submittals by the Engineer's approval thereof.
- 3.11.9 The Contractor shall direct specific attention, in writing or on resubmitted shop drawings,

product data, samples, or similar submittals, to revisions other than those requested by the Engineer on previous submittals.

- 3.11.10 Informational submittals upon which the Engineer is not expected to take responsive action may be so identified in the contract documents.
- 3.11.11 When professional certification of performance criteria of materials, systems, or equipment is required by the contract documents, the Engineer shall be entitled to rely upon the accuracy and completeness of such calculations and certifications.
- 3.11.12 All shop drawings for any structural, mechanical, or electrical work shall be submitted to, and approved by, the Engineer.

3.12 USE OF SITE

- 3.12.1 The Contractor shall confine operations at the site to areas permitted by law, ordinances, permits, and the contract documents and shall not unreasonably encumber the site with materials or equipment.
- 3.12.2 Protection of construction materials and equipment stored at the project site from weather, theft, or damage is solely the responsibility of the Contractor.

3.13 CUTTING AND PATCHING

- 3.13.1 The Contractor shall be responsible for cutting, fitting, or patching required to complete the work or to make its parts fit together properly.
- 3.13.2 The Contractor shall not damage or endanger a portion of the work or fully or partially completed construction of the Owner Representative or separate Contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner Representative or a separate Contractor except with written consent of the Owner Representative and of such separate Contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner Representative or a separate Contractor the Contractor's consent to cutting or otherwise altering the work.

3.14 CLEAN UP

- 3.14.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the contract. At completion of the work the Contractor shall remove from and about the project waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials.
- 3.14.2 If the Contractor fails to clean up as provided in the contract documents, the Owner Representative may do so and the cost thereof shall be charged to the Contractor.

3.15 ACCESS TO WORK AND DOCUMENTS

3.15.1 The Contractor shall provide the Owner Representative and Engineer access to the work in

preparation and progress wherever located and any books, documents, papers, and records of the Contractor which are directly pertinent to the contract for the purpose of making audit, examination, excerpts, and transcriptions.

3.16 ROYALTIES AND PATENTS

3.16.1 The Contractor shall pay all royalties and license fees. The Contractor shall defend suits, or claims for infringement of patent rights and shall hold the Owner Representative, the State of Alaska, its agents and employees harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process, or product of a particular manufacturer or manufacturers is required by the contract documents. However, if the Contractor has reason to believe that the required design, process, or product is an infringement of a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Engineer.

3.17 INDEMNIFICATION

- 3.17.1 To the fullest extent permitted by law, the Contractor shall defend, indemnify and hold harmless the Owner, Owner Representative, the State of Alaska, its agents and employees from and against claims, damages, losses, and expenses, including but not limited to attorney's fees, arising out of or resulting from performance of the work, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease, or death, or to injury to or destruction of tangible property (other than the work itself) including loss of use resulting therefrom, but only to the extent caused in whole or in part by negligent acts or omissions of the Contractor, a subcontractor of whatever tier, anyone directly or indirectly employed by them or anyone for whose acts they may be liable. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to a party or person described in this paragraph.
- 3.17.2 In claims against any person or entity indemnified under paragraph 3.17 by an employee of the Contractor, a subcontractor of any tier, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation shall not be limited by a limitation on amount or type of damages, compensation or benefits payable under workers' compensation acts, disability benefit acts or other employee benefit acts.
- 3.17.3 The obligations of the Contractor under paragraph 3.17 shall not extend to the liability of the Engineer, the Engineer's consultants, and agents and employees of any of them arising out of (1) the preparation or approval of maps, drawings, opinions, reports, surveys, change orders, designs or specifications, or (2) the giving of or the failure to give directions or instructions by the Engineer, the Engineer's consultants, and agents and employees of any of them.

3.18 BUSINESS AND PROFESSIONAL REGISTRATION

- 3.18.1 The Contractor shall comply and ensure that its employees and subcontractors of every tier comply with all applicable laws and ordinances related to Alaska business and professional licensing.
- 3.18.2 The Contractor shall provide copies of licenses within seven days following a request from the Owner Representative's representative for the duration of this agreement.

3.19 PROJECT RECORDS

3.19.1 The Contractor shall maintain all records relating to the work for a period of six years from the date of final completion and shall, upon request, make such records available for inspection by Owner Representative, or the State of Alaska during regular business hours upon reasonable advance notice.

ARTICLE 4

ADMINISTRATION OF THE CONTRACT

4.1 ENGINEER

- 4.1.1 Any reference in the contract documents to the Engineer taking action or rendering a decision within a "reasonable time" is understood to mean no more than 14 days.
- 4.1.2 Duties, responsibilities, and limitations on the authority of the Engineer as set forth in the contract documents shall not be restricted, modified, or extended without the written consent of the Owner Representative.

4.2 ENGINEER'S ADMINISTRATION OF THE CONTRACT

- 4.2.1 The Engineer shall provide administration of the contract as described in the contract documents, and shall represent the Owner Representative's during construction of the work at the Owner Representative's direction. The Engineer will advise and consult with the Owner Representative. The Engineer will have authority to act on behalf of the Owner Representative only to the extent provided in the contract documents, unless otherwise provided in a written agreement signed by the Owner Representative.
- 4.2.2 The Engineer shall visit the site as requested by the Owner Representative to determine if the work is being performed in a manner indicating that the work when completed will be in accordance with the design and specifications for the project. The Engineer will coordinate with the Contractor's qualified inspector and superintendent for these periodic reviews.
- 4.2.3 The Engineer will not have control over or charge of and will not be responsible for construction means, methods, techniques, sequences, or procedures, or for safety precautions and programs in connection with the work. Except as otherwise provided in the contract documents, the Engineer will not be responsible for the Contractor's failure to carry out the work in accordance with the contract documents. Except as otherwise provided in the contract documents, the Engineer will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, subcontractors, or their agents or employees, or of any other persons performing portions of the work.
- 4.2.4 Communications Facilitating Contract Administration. Except as otherwise provided in the contract documents, the Owner Representative and Contractor may endeavor to communicate through the Engineer. Communications by and with the Engineer's consultants shall be through the Engineer. Communications by and with subcontractors of any tier and material suppliers shall be through the Contractor. Communications by and with separate Contractors shall be through the Owner Representative.

- 4.2.5 Based on the Engineer's observations at the site of the work and evaluations of the Contractor's applications for payment, the Engineer shall review and certify the amounts due the Contractor and shall issue certificates for payment in such amounts.
- 4.2.6 The Engineer will have authority to reject work which does not conform to the contract documents. Whenever the Engineer considers it necessary or advisable for implementation of the intent of the contract documents, the Engineer will have authority to require additional inspection or testing of the work in accordance with subparagraphs 13.5.2 and 13.5.3, whether or not such work is fabricated, installed, or completed.
- 4.2.7 The Engineer shall review and approve or take other appropriate action upon the Contractor's submittals such as shop drawings, product data, and samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the contract documents. The Engineer's action will be taken with such reasonable promptness as to cause no delay in the work or in the activities of the Owner Representative, Contractor or separate Contractors. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the contract documents. The Engineer's review of the Contractor's submittals shall not relieve the Contractor of the obligations under paragraphs 3.3, 3.5, and 3.11. The Engineer's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Engineer, of any construction means, methods, techniques, sequences or procedures. The Engineer's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- 4.2.8 Change orders and construction change directives with supporting documentation shall be completed by the Engineer upon approval and authorization from the Owner's Representative in accordance with Paragraphs 7.2, 7.3, and 7.4.
- 4.2.9 On behalf of the Owner Representative, the Engineer shall conduct inspections to determine the dates of substantial completion and final completion and shall issue a certificate of substantial completion. The Engineer will receive and review (and approve or disapprove, as the case may be) written guarantees, and related documents required by the contract for construction to be assembled by the Contractor, and shall issue a final certificate of payment upon compliance with the requirements of the contract documents. The Engineer shall conduct a one year warranty inspection to determine if warranty work is needed or completed.
- 4.2.10 If the Owner Representative and Engineer agree, the Engineer will provide one or more project representatives to assist in carrying out the Engineer's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in the agreement between the Owner Representative and the Engineer.
- 4.2.11 The Engineer may observe the construction at certain key periods of completion, in addition to other observations. The Contractor shall notify the Engineer five days in advance of commencing work as listed below, to allow the Engineer to schedule observations of the following, if applicable:
 - Underground utilities prior to cover, including sewer, water, storm sewer, and electrical.
 - Foundation reinforcing prior to concrete placement.
 - Mechanical and electrical rough-in work prior to cover.

- Substantial completion.
- Final completion.

4.3 CLAIMS AND DISPUTES

- 4.3.1 <u>Definition.</u> A "claim" is a demand or assertion by the Owner Representative or Contractor seeking, as a matter of right, adjustment or interpretation of contract terms, payment of money, extension of time or other relief with respect to the terms of the contract. The term "claim" also includes other disputes and matters in question between the Owner Representative and Contractor arising out of or relating to the contract. Claims must be made by written notice. The responsibility to substantiate claims shall rest with the party making the claim.
- 4.3.2 <u>Time Limits on Claims.</u> Claims by either party must be made within 21 days after occurrence of the event giving rise to such claim. Claims must be made by written notice. An additional claim made after the initial claim has been implemented by change order will not be considered unless submitted in a timely manner. Any notice of claim must clearly identify the alleged cause and the nature of the claim and include data and information then available to the claimant that will facilitate proper verification and evaluation of the claim.
- 4.3.3 <u>Continuing Contract Performance.</u> Pending final resolution of a claim, unless otherwise agreed in writing the Contractor shall proceed diligently with performance of the contract and the Owner Representative shall continue to make payments in accordance with the contract documents.
- 4.3.4 <u>Waiver of Claims and Final Payment.</u> The making of final payment shall constitute a waiver of claims by the Owner Representative except those arising from:
 - 4.3.4.1 Liens, claims, security interests or encumbrances arising out of the contract and unsettled;
 - 4.3.4.2 Failure of the work to comply with the requirements of the contract documents; or
 - 4.3.4.3 Terms of special warranties required by the contract documents.
- 4.3.5 Claims for Concealed or Unknown Conditions. If conditions are encountered at the site which are (1) subsurface or otherwise concealed physical conditions which differ materially from those indicated in the contract documents, or (2) unknown physical conditions of an unusual nature, which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the contract documents, or (3) hazardous materials at the site contain constituents, or are present in quantities not disclosed in available information, or they have characteristics or properties not disclosed by such information, and such constituents, characteristics, properties, or quantities increase the risk of hazard to human health or the environment involved in the performance of the work under this agreement, then notice by the observing party shall be given to the other party promptly before conditions are disturbed and in no event later than two days after first observance of the conditions. The Engineer will promptly investigate such conditions and, if they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the work, will recommend an equitable adjustment in the contract sum or contract time, or both. If the Engineer determines that

the conditions at the site are not materially different from those indicated in the contract documents and that no change in the terms of the contract is justified, the Engineer shall so notify the Owner Representative and Contractor in writing, stating the reasons. Claims by either party in opposition to such determination must be made within 21 days after the Engineer has given notice of decision under this subparagraph. If the Owner Representative and Contractor cannot agree on an adjustment in the contact sum or contract time, the adjustment shall be referred to the Engineer for initial determination, subject to further proceedings pursuant to paragraph 4.4. No adjustment in the contract time or contract sum shall be permitted, however, in connection with a concealed or unknown condition which does not differ materially from those conditions disclosed or which reasonably should have been disclosed by the Contractor's prior inspections, tests, reviews, and pre-bid investigation for the project.

- 4.3.6 <u>Claims for Additional Cost.</u> If the Contractor wishes to make claim for an increase in the contract sum, written notice as provided herein shall be given before proceeding to execute the proposed additional work. Prior notice is not required for claims relating to an emergency endangering life or property arising under paragraph 10.3.
- 4.3.7 If the Contractor believes additional costs may be incurred for reasons including but not limited to (1) a written interpretation from the Engineer, (2) an order by the Owner Representative to stop the work where the Contractor was not at fault, (3) a written order for a minor change in the work issued by the Engineer, (4) failure of payment by the Owner Representative, (5) termination of the contract by the Owner Representative, (6) Owner Representative's suspension or, (7) other grounds, claim shall be filed in accordance with the procedure established herein.
- 4.3.8 <u>Claims for Additional Time</u> If the Contractor wishes to make claim for an increase in the contract time, written notice as provided herein shall be given. The Contractor's claim shall include an estimate of cost and of probable effect of delay on progress of the work. In the case of a continuing delay only one claim is necessary. If adverse weather conditions are the basis for a claim for additional time, such claim shall be documented by data substantiating that weather conditions were highly unusual for the period of time and could not have been reasonably anticipated, and that weather conditions had an adverse effect on the scheduled construction. No claim shall be allowed for weather conditions that were not highly unusual.
- 4.3.9 <u>Injury or Damage to Person or Property.</u> If either party to the contract suffers injury or damage to person or property because of an act or omission of the other party, of any of the other party's employees or agents, or of others for whose acts such party is legally liable, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after first observance. The notice shall provide sufficient detail to enable the other party to investigate the matter. If a claim for additional cost or time is to be asserted as a result, it shall be filed as provided in subparagraphs 4.3.6 through 4.3.8.
- 4.3.10 If unit prices are stated in the contract documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed change order or construction change directive so that application of such unit prices to quantities or work proposed will cause substantial inequity to the Owner Representative or Contractor, the applicable unit prices shall be equitably adjusted.

4.4 RESOLUTION OF CLAIMS AND DISPUTES

- 4.4.1 <u>Decision of Engineer.</u> Claims shall be referred initially to the Engineer for action as provided in paragraph 4.4 if the claimant first recognizes the claim prior to the date of final payment. A decision by the Engineer, as provided in subparagraph 4.4.3, shall be required as a condition precedent to litigation of a claim between the Contractor and Owner Representative as to all such matters arising prior to the date final payment is due, regardless of (1) whether such matters relate to execution and progress of the work, or (2) the extent to which the work has been completed. The decision by the Engineer in response to a claim shall not be a condition precedent to litigation in the event (1) the position of Engineer is vacant, (2) the Engineer has not received evidence or has failed to render a decision within agreed time limits, (3) the Engineer has failed to take action required under subparagraph 4.4.3 within 30 days after the claim is made, or (4) 45 days have passed after the claim has been referred to the Engineer.
- 4.4.2 The Engineer will review claims and take one or more of the following preliminary actions within 10 days of receipt of a claim: (1) request additional supporting data from the claimant, (2) submit a schedule to the parties indicating when the Engineer expects to take action, (3) reject the claim in whole or in part, stating reasons for rejection, (4) recommend approval of the claim by the other party, or (5) suggest a compromise. The Engineer may also, but is not obligated to, notify the surety, if any, of the nature and amount of the claim. The Engineer must notify the Village Safe Water Program and receive approval before rendering a decision regarding a Contractor's claim.
- 4.4.3 If the Engineer requests a party to provide a response to a claim or to furnish additional supporting data, such party shall respond, within 10 days after receipt of such request, and shall either provide a response or the requested supporting data, advise the Engineer when the response or supporting data will be furnished or advise the Engineer that no supporting data will be furnished. Upon receipt of the response, or supporting data, if any, the Engineer will either reject or approve the claim in whole or in part.
- 4.4.4 If a claim has not been resolved after consideration of the foregoing and of further evidence presented by the parties or requested by the Engineer, the Engineer will notify the parties in writing that the Engineer's decision will be made within seven days. Upon expiration of such time period, the Engineer will render to the parties the Engineer's written decision relative to the claim, including any recommended change in the contract sum or contract time or both. If there is a surety and there appears to be a possibility of a Contractor's default, the Engineer may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy. The Engineer must notify the Village Safe Water Program and receive approval before rendering a decision regarding a Contractor's claim.

4.5 VENUE

4.5.1 Any lawsuit arising out of or in any way related to this project shall be exclusively brought and maintained in the Superior Court for the State of Alaska.

ARTICLE 5

SUBCONTRACTORS

5.1 **DEFINITIONS**

5.1.1 The term "subcontractor" is referred to throughout the contract documents as if singular in number and means a subcontractor or an authorized representative of the subcontractor. The term "subcontractor" does not include a separate Contractor or subcontractors of a separate Contractor.

5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- 5.2.1 Within five working days after the identification of the apparent low bidder for a construction contract, the apparent low bidder shall submit a list of the subcontractors the bidder proposes to use in the performance of the construction contract. The list must include the name and location of the place of business for each subcontractor, evidence of each subcontractor's valid Alaska business license, and evidence of each subcontractor's registration as a contractor under AS 08.18. If a subcontractor on the list did not have a valid Alaska business license and a valid certificate of registration as a Contractor under AS 08.18 at the time the bid was opened, the bidder may not use the subcontractor in the performance of the contract, and shall replace the subcontractor with a subcontractor who had a valid Alaska business license and a valid certificate of registration as a Contractor under AS 08.18 at the time the bid was opened.
- 5.2.2 The Contractor shall not allow a proposed person or entity to whom the Owner Representative has made reasonable objection to perform any work on the project.
- 5.2.3 The Contractor shall not change a subcontractor, person or entity previously selected if the Owner Representative makes reasonable objection to such change.

5.3 SUBCONTRACTUAL RELATIONS

By appropriate agreement, written where legally required for validity, the Contractor shall require each subcontractor, to the extent of the work to be performed by the subcontractor, to be bound to the Contractor by terms of the contract documents, and to assume toward the Contractor all the obligations and responsibilities which the Contractor, by these documents, assumes toward the Owner Representative and Engineer. Each subcontract agreement shall preserve and protect the rights of the Owner Representative and Engineer under the contract documents with respect to the work to be performed by the subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the contract documents, has against the Owner Representative. Where appropriate, the Contractor shall require each subcontractor of whatever tier to enter into similar agreements with their subcontractors. The Contractor shall make available to each proposed subcontractor, prior to the execution of the subcontract agreement, copies of the contract documents to which the subcontractor will be bound, and, upon written request of the subcontractor, identify to the subcontractor terms and conditions of the proposed subcontract agreement which may be at variance with the contract documents. Subcontractors of every tier shall similarly make copies of applicable portions of such documents

available to their respective proposed subcontractors.

5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

- 5.4.1 Each subcontract agreement for a portion of the work is assigned by the Contractor to the Owner Representative provided that:
 - 5.4.1.1 Assignment is effective only after termination of the contract by the Owner Representative for cause pursuant to paragraph 14.1 and only for those subcontract agreements which the Owner Representative accepts by notifying the subcontractor in writing; and
 - 5.4.1.2 Assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the contract.
- 5.4.2 Each subcontract shall specifically provide that the Owner Representative shall only be responsible to the subcontractor for those obligations of the Contractor that accrue subsequent to the Owner Representative's exercise of any rights under this conditional assignment.

ARTICLE 6

CONSTRUCTION BY OWNER REPRESENTATIVE OR BY SEPARATE CONTRACTORS

6.1 OWNER REPRESENTATIVE'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

- 6.1.1 The Owner Representative reserves the right to perform construction or operations related to the project with the Owner Representative's own forces, and to award separate contracts in connection with other portions of the project or other construction or operations on the site under conditions of the contract identical or substantially similar to these including those portions related to insurance and wavier of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner Representative, the Contractor shall make such claim as provided elsewhere in the contract documents.
- 6.1.2 When separate contracts are awarded for different portions of the project, or other construction, or operations on the site, the term "Contractor" in the contract documents in each case shall mean the Contractor who executes each separate Owner Representative-Contractor agreement.
- 6.1.3 The Owner Representative shall provide for coordination of the activities of the Owner Representative's own forces and of each separate Contractor with the work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate Contractors and the Owner Representative in reviewing their construction schedules when directed to do so.
- 6.1.4 Unless otherwise provided in the contract documents, when the Owner Representative performs construction or operations related to the project with the Owner Representative's own forces, the Owner Representative shall be deemed to be subject to the same obligations and to have the same

rights which apply to the Contractor under the conditions of the contract, including, without excluding others, those stated in Article 3, 6, 10, 11, and 12.

6.2 MUTUAL RESPONSIBILITY

- 6.2.1 The Contractor shall afford the Owner Representative and separate Contractor's reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities and shall connect and coordinate the Contractor's construction and operations with theirs as required by the contract documents.
- 6.2.2 If part of the Contractor's work depends for proper execution or results upon construction or operations by the Owner Representative or a separate Contractor, the Contractor shall, prior to proceeding with that portion of the work, promptly report to the Engineer apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner Representative's or separate Contractors' completed or partially completed construction is fit and proper to receive the Contractor's work, except as to defects not then reasonably discoverable.
- 6.2.3 Costs caused by delays, or by improperly timed activities, or defective construction shall be borne by the party responsible therefor.
- 6.2.4 The Contractor shall promptly remedy damage wrongfully caused by the Contractor to complete or partially completed construction, or to property of the Owner Representative, or separate Contractors as provided in subparagraph 10.2.4.
- 6.2.5 The Owner Representative and each separate Contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in paragraph 3.13.
- 6.2.6 Claims and other disputes and matters in question between the Contractor and a separate Contractor shall be subject to the provisions of paragraph 4.3 provided the separate Contractor has reciprocal obligations.

ARTICLE 7

CHANGES IN THE WORK

7.1 CHANGES

7.1.1 Changes in the work may be accomplished after execution of the contract, and without invalidating the contract, by change order, construction change directive, or order for a minor change in the work, subject to the limitations stated in this Article 7 and elsewhere in the contract documents.

- 7.1.2 A change order shall be based upon agreement among the Owner Representative and Contractor; a construction change directive is a direction from the Owner Representative and may or may not be agreed to by the Contractor; an order for a minor change in the work may be issued by the Engineer alone.
- 7.1.3 Changes in the work shall be performed under applicable provisions of the contract documents, and the Contractor shall proceed promptly, unless otherwise provided in the change order, construction change directive, or order for a minor change in the work. Except as permitted in paragraph 7.3, a change in the contract sum or the contract time shall be accomplished only by change order. Accordingly, no course of conduct or dealings between the parties, nor express or implied acceptance of alterations or additions to the work, nor any claim that the Owner Representative has been unjustly enriched by any alteration or addition to the work, shall be the basis of any claim to an increase in any amount due under the contract documents or a change in any time period provided for in the contract documents.

7.2 CHANGE ORDERS

- 7.2.1 A change order is a written instrument prepared by the Engineer and signed by the Owner Representative, Contractor and Engineer, stating their agreement upon all of the following:
 - 7.2.1.1 A change in the work;
 - 7.2.1.2 The amount of the adjustment in the contract sum, if any; and
 - 7.2.1.3 The extent of the adjustment in the contract time, if any.
- 7.2.2 Methods used in determining adjustment to the contract sum may include those listed in subparagraph 7.3.3.
- 7.2.3 Agreement on any change order shall constitute a final settlement of all matters relating to the change in the work which is the subject of the change order, including, but not limited to, all direct and indirect costs associated with such change and any and all adjustments to the contract sum and the construction schedule. The one year warranty will apply on approved change orders.

7.3 CONSTRUCTION CHANGE DIRECTIVES

- 7.3.1 A construction change directive is a written order prepared by the Engineer and signed by the Owner Representative and Engineer, directing a change in the work and stating a proposed basis for adjustment, if any, in the contract sum or contract time, or both. The Owner Representative may by construction change directive, without invalidating the contract, order changes in the work within the general scope of the contract consisting of additions, deletions or other revisions, the contract sum and contract time being adjusted accordingly.
- 7.3.2 A construction change directive shall be used in the absence of total agreement on the terms of a change order.

- 7.3.3 If the construction change directive provides for an adjustment to the contract sum, the adjustment shall be based on one of the following methods:
 - 7.3.3.1 mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
 - 7.3.3.2 unit prices stated in the contract documents or subsequently agreed upon;
 - 7.3.3.3 cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
 - 7.3.3.4 as provided in subparagraph 7.3.6.
- 7.3.4 Upon receipt of a construction change directive, the Contractor shall promptly proceed with the change in the work involved and advise the Engineer of the Contractor's agreement or disagreement with the method, if any, provided in the construction change directive for determining the proposed adjustment in the contract sum or contract time.
- 7.3.5 A construction change directive signed by the Contractor indicates the agreement of the Contractor therewith, including adjustment in contract sum and contract time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a change order.
- 7.3.6 If the Contractor does not execute the construction change directive within 10 days or disagrees with the method for adjustment in the contact sum, the method and the adjustment shall be determined by the Owner Representative on the basis of reasonable expenditures and savings of those performing the work attributable to the change, including, in case of an increase in the contract sum, a reasonable allowance for overhead and profit. In such case, and also under clause 7.3.3.3, the Contractor shall keep and present, in such form as the Engineer may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the contract documents, costs for the purposes of this subparagraph 7.3.6 shall be limited to the following:
 - 7.3.6.1 costs of labor, including social security, unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;
 - 7.3.6.2 costs of equipment, materials, and supplies, including cost of transportation, whether incorporated or consumed;
 - 7.3.6.3 <u>Rental Costs of Machinery and Equipment.</u> The rental value of the Contractor's own equipment shall be not more than 100% of the rates in the current edition of Equipment Watch Rental Rate Blue Book for Construction Equipment, and the aggregate amounts charged to the Owner Representative for such equipment shall not exceed 50% of its fair market value;
 - 7.3.6.4 costs of premiums for all bond and insurance, permit fees, and sales, use or similar taxes related to the work; and

- 7.3.6.5 additional costs of supervision and field office personnel directly attributable to the change.
- 7.3.7 Pending final determination of cost to the Owner Representative, amounts not in dispute may be included in applications for payment. The amount of credit to be allowed by the Contractor to the Owner Representative for a deletion or change which results in a net decrease in the contract sum shall be actual net cost as confirmed by the Engineer. When both additions and credits covering related work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- 7.3.8 If the Owner Representative and Contractor do not agree with the adjustment in contract time or the method for determining it, the adjustment or the method shall be referred to the Engineer for recommendation.
- 7.3.9 When the Owner Representative and Contractor agree with the recommendation made by the Engineer concerning the adjustments in the contract sum and contract time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and shall be recorded by preparation and execution of an appropriate change order.

7.4 MINOR CHANGES IN THE WORK

7.4.1 The Engineer may order minor changes in the work not involving adjustment in the contract sum or extension of the contract time and not inconsistent with the intent of the contract documents. Such changes shall be effected by written order and shall be binding on the Owner Representative and Contractor. The Contractor shall carry out such written orders promptly.

ARTICLE 8

TIME

8.1 **DEFINITIONS**

- 8.1.1 Unless otherwise provided, contract time is the period of time, including authorized adjustments, allotted in the contract documents for substantial completion of the work.
- 8.1.2 The date of commencement of the work is the date established in the agreement. The date shall not be postponed by the failure to act of the Contractor or of persons or entities for whom the Contractor is responsible.
- 8.1.3 The date of substantial completion is the date certified by the Engineer in accordance with paragraph 9.7.
- 8.1.4 The term "day" as used in the contract documents shall mean calendar day unless otherwise specifically defined.

8.2 PROGRESS AND COMPLETION

- 8.2.1 Time limits stated in the contract documents are of the essence of the contract. By executing the agreement the Contractor confirms that the contract time is a reasonable period for performing the work.
- 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner Representative in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor. The date of commencement of the work shall not be changed by the effective date of such insurance.
- 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve substantial completion within the contract time.

8.3 DELAYS AND EXTENSIONS OF TIME

- 8.3.1 If the Contractor is delayed at any time in progress of the work by an act or neglect of the Owner Representative or Engineer, or of an employee of either, or of a separate Contractor employed by the Owner Representative, or by changes ordered in the work, or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control, or by delay authorized by the Owner Representative, then the contract time shall be extended by change order to the extent such delay will prevent the Contractor from achieving substantial completion within the contract time and if the performance of the work is not, was not, or would not have been delayed by any other cause for which the Contractor is not entitled to an extension in the contract time under the contract documents. Adjustments in the contract time shall be permitted for a delay only to the extent such delay is not caused, or could not have been anticipated by the Contractor, could not be limited or avoided by the Contractor's timely notice to the Owner Representative of the delay, and is of a duration of at least one day.
- 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of paragraph 4.3.
- 8.3.3 This paragraph 8.3 does not preclude recovery of damages for delay by either party under other provisions of the contract documents. However, in no event shall the Contractor be entitled to recovery of consequential damages, lost opportunity costs, unabsorbed home office overhead, or impact damages. The Contractor's sole remedy for delay in commencement, prosecution or completion of the work, hindrance or obstruction in the performance of the work, loss of productivity or similar claims is an extension of time in which to complete the work if permitted under the contract documents and, to the extent permitted under this paragraph, an adjustment in the contract sum. Any adjustment in the contract sum pursuant to this paragraph shall be limited to the increase, if any, of direct, out of pocket, on-site labor and material costs incurred by the Contractor in performing the work as a result of that portion of any delay or delays which cause the contract time to be increased. Such direct costs shall not include profit or overhead.

PAYMENTS AND COMPLETION

9.1 CONTRACT SUM

9.1.1 The contract sum is stated in the agreement and, including authorized adjustments, is the total amount payable by the Owner Representative to the Contractor for performance of the work under the contract documents.

9.2 SCHEDULE OF VALUES

9.2.1 Before the first application for payment, the Contractor shall submit to the Engineer a schedule of values allocated to various portions of the work, prepared in such form and supported by such data to substantiate its accuracy as the Engineer may require. This schedule, unless objected to by the Engineer, shall be used as a basis for reviewing the Contractor's applications for payment.

9.3 APPLICATIONS FOR PAYMENT

- 9.3.1 The Contractor shall submit to the Engineer an itemized application for payment for operations completed in accordance with the schedule of values. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner Representative or Engineer may require, such as copies of requisitions from subcontractors and material suppliers, and reflecting retainage if provided for elsewhere in the contract documents.
 - 9.3.1.1 Such applications may include requests for payment on account of changes in the work which have been properly authorized by construction change directives but not yet included in change orders.
 - 9.3.1.2 Such applications may not include requests for payment of amounts the Contractor does not intend to pay to a subcontractor or material supplier because of a dispute or other reason.
- 9.3.2 The Contractor warrants that title to all work covered by an application for payment will pass to the Owner Representative no later than the time of payment. The Contractor further warrants that upon submittal of an application for payment, all work for which certificates for payment have been previously issued and payments received from the Owner Representative shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests, or encumbrances in favor of the Contractor, subcontractors of any tier, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the work.

9.4 CERTIFICATES FOR PAYMENT

- 9.4.1 The Engineer will, within seven days after receipt of the Contractor's application for payment, either issue to the Owner Representative a certificate for payment, with a copy to the Contractor, for such amount as the Engineer believes is properly due, or notify the Contractor and Owner Representative in writing of the Engineer's reasons for withholding certification in whole or in part as provided in subparagraph 9.5.1.
- 9.4.2 The issuance of a certificate for payment will constitute a representation by the Engineer to the Owner Representative, based on the Engineer's observations at the site and the data comprising the application for payment, that the work has progressed to the point indicated and that, to the best of the Engineer's knowledge, information and belief, quality of the work is in accordance with the contract documents. The foregoing representations are subject to minor deviations from the contract documents correctable prior to completion and to specific qualifications expressed by the Engineer. The issuance of a certificate for payment will further constitute a representation that the Contractor is entitled to payment in the amount certified.

9.5 DECISIONS TO WITHHOLD CERTIFICATION

- 9.5.1 The Engineer shall decide not to certify payment and shall withhold a certificate for payment in whole or in part, to the extent reasonably necessary to protect the Owner Representative, if in the Engineer's opinion the representations to the Owner Representative required by subparagraph 9.4.2 cannot be made. If the Engineer is unable to certify payment in the amount of the application, the Engineer shall notify the Contractor and Owner Representative as provided in subparagraph 9.4.1. If the Contractor and Engineer cannot agree on a revised amount, the Engineer shall promptly issue a certificate for payment for the amount for which the Engineer is able to make such representations to the Owner Representative. The Engineer shall also decide not to certify payment or, because of subsequently discovered evidence or subsequent observations, shall nullify the whole or a part of a certificate for payment previously issued, to such extent as may be necessary in the Engineer's opinion to protect the Owner Representative from loss for which Contractor is responsible, including loss resulting from acts and omissions described in subparagraph 3.3.2, because of:
 - 9.5.1.1 defective work not remedied;
 - 9.5.1.2 third party claims filed or reasonable evidence indicating probable filing of such claims;
 - 9.5.1.3 failure of the Contractor to make payments properly to subcontractors or for labor, materials or equipment;
 - 9.5.1.4 reasonable evidence that the work cannot be completed for the unpaid balance of the contract sum;
 - 9.5.1.5 damage to the Owner Representative or another Contractor;
 - 9.5.1.6 reasonable evidence that the work will not be completed within the contract time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the

anticipated delay; or

- 9.5.1.7 persistent failure to carry out the work in accordance with the contract documents.
- 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

9.6 PROGRESS PAYMENTS

- 9.6.1 After the Engineer has issued a certificate for payment, the Owner Representative shall make payment in the manner and within the time provided in the contract documents, and shall so notify the Engineer.
- 9.6.2 The Contractor shall promptly pay each subcontractor, upon receipt of payment from the Owner Representative, out of the amount paid to the Contractor on account of such subcontractor's portion of the work, the amount to which said subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of such subcontractor's portion of the work. The Contractor shall, by appropriate agreement with each subcontractor, require each subcontractor to make payments to sub-subcontractors in similar manner. Owner Representative may elect, in Owner Representative's sole discretion, to make any payment requested by the Contractor on behalf of a subcontractor of any tier jointly payable to the Contractor and such subcontractor. The Contractor and subcontractor shall be responsible for the allocation and disbursement of funds included as part of any such joint payment. In no event shall any joint payment be construed as (1) a contract between the Owner Representative and subcontractor of any tier, (2) creating any obligations to such subcontractor on the part of Owner Representative, or (3) creating any rights in such subcontractor against the Owner Representative.
- 9.6.3 The Engineer will, on request, furnish to a subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Engineer and Owner Representative on account of portions of the work done by such subcontractor.
- 9.6.4 Neither the Owner Representative nor Engineer shall have an obligation to pay or provide oversight of payments of money to a subcontractor except as may otherwise be required by law.
- 9.6.5 Payment to material suppliers shall be treated in a manner similar to that provided in subparagraphs 9.6.2, 9.6.3 and 9.6.4.
- 9.6.6 A certificate for payment, a progress payment, or partial or entire use or occupancy of the project by the Owner Representative shall not constitute acceptance of work not in accordance with the contract documents.
- 9.6.7 If the Owner Representative is entitled to reimbursement or payment from the Contractor pursuant to the contract documents, such payment shall be made promptly upon demand by the Owner Representative. If Contractor fails to promptly make any payment due the Owner Representative, or the Owner Representative incurs any costs and expenses to cure any default of the Contractor or to correct defective work, the Owner Representative shall have an absolute right to offset such amount against the contract sum and may, in the Owner Representative's sole discretion, elect either to: (1)

deduct an amount equal to that which the Owner Representative is entitled from any payment then or thereafter due the Contractor from the Owner Representative; or (2) issue a written notice to the Contractor reducing the contract sum by an amount equal to that which the Owner Representative is entitled.

9.7 SUBSTANTIAL COMPLETION

- 9.7.1 Substantial completion is the stage in the progress of the work when the work or designated portion thereof is sufficiently complete in accordance with the contract documents so the Owner Representative can occupy or utilize the work for its intended use.
- 9.7.2 When the Contractor considers that the work, or a portion thereof which the Owner Representative agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Engineer a comprehensive list of items to be completed or corrected. The Contractor shall proceed promptly to complete and correct items on the list. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all work in accordance with the contract documents.
- 9.7.3 Upon receipt of the Contractor's list, the Engineer shall make an inspection to determine whether the work or designated portion thereof is substantially complete. If the Engineer's inspection discloses any item, whether or not included on the Contractor's list, which is not in accordance with the requirements of the contract documents, the Contractor shall, before issuance of the certificate of substantial completion, complete or correct such item upon notification by the Engineer. The Contractor shall then submit a request for another inspection by the Engineer to determine substantial completion.
- 9.7.4 When the work or designated portion thereof is substantially complete, the Engineer shall prepare a certificate of substantial completion which shall establish the date of substantial completion, shall establish responsibilities of the Owner Representative and Contractor for security, maintenance, heat, utilities, damage to the work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the certificate which shall identify all non-conforming, defective and incomplete work and establish the date of commencement of warranties in connection with any such work. Warranties required by the contract documents shall commence on the date of substantial completion of the work or designated portion thereof unless otherwise provided in the certificate of substantial completion.
- 9.7.5 The certificate of substantial completion shall be submitted to the Owner Representative and Contractor for their written acceptance of responsibilities assigned to them in such certificate. Upon substantial completion of the work or designated portion thereof, and upon application by the Contractor and certification by the Engineer, the Owner Representative shall make payment, reflecting adjustment in retainage, if any, for such work or portion thereof as provided in the contract documents.

9.8 FINAL COMPLETION AND FINAL PAYMENT

9.8.1 Upon receipt of written notice that the work is ready for final inspection and acceptance, and upon receipt of a final application for payment, the Engineer shall promptly make such inspection and, when the Engineer finds the work acceptable under the contract documents and the contract fully

performed, the Engineer shall promptly issue a final certificate for payment stating that to the best of the Engineer's knowledge, information and belief, and on the basis of the Engineer's observations and inspections, the work has been completed in accordance with the terms and conditions of the contract documents and that the entire balance found to be due the Contractor and noted in said final certificate is due and payable. The Engineer's final certificate for payment shall constitute a further representation that conditions listed in subparagraph 9.8.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

9.8.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Engineer: (1) an affidavit that includes payrolls, bills for materials and equipment, and other indebtedness connected with the work for which the Owner or the Owner Representative's property might be responsible or encumbered (less amounts withheld by Owner Representative) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the contract documents to remain in force after final payment is currently in effect and will not be cancelled or allowed to expire until at least 30 days' prior written notice has been given to the Owner Representative, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the contract documents (4) consent of surety, if any, to final payment, (5) verification from the State of Alaska Department of Labor and Workforce Development ("the Department") that: (i) Contractor has complied with AS 36.05.045(a) and (b); (ii) the Department is not conducting an investigation under Title 36 of the Alaska Statutes and (iii) the Department has not issued a notice of violation of AS 36.05 to Contractor or any subcontractor working on the project, and (6), if required by the Owner Representative, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the contract, to the extent and in such form as may be designated by the Owner Representative. If a subcontractor refuses to furnish a release or waiver required by the Owner Representative, the Contractor may furnish a bond satisfactory to the Owner Representative to indemnify the Owner Representative against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner Representative all money that the Owner Representative may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

If, after substantial completion of the work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of change orders affecting final completion, and the Engineer so confirms, the Owner Representative shall, upon application by the Contractor and certification by the Engineer, and without terminating the contract, make payment of the balance due for that portion of the work fully completed and accepted. If the remaining balance for work not fully completed or corrected is less than retainage stipulated in the contract documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the work fully completed and accepted shall be submitted by the Contractor to the Engineer prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims. The making of final payment shall constitute a waiver of claims by the Owner Representative as provided in subparagraph 4.3.5. Acceptance of final payment by the Contractor, a subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final application for payment. Such waivers shall be in addition to the waiver described in subparagraph 4.3.5. Owner Representative shall withhold from final payment an amount sufficient to pay the department's estimate of what may be needed to compensate the employees of Contractor or

any subcontractor under investigation pursuant to AS 36.05 plus any unpaid AS 36.05.045 filing fee.

9.9 **RETAINAGE**

9.9.1 At any time the Owner Representative finds that satisfactory progress is not being made it may retain a maximum amount equal to 10% of the total amount earned on all subsequent progress payments. This retainage may be released at such time as the Owner Representative finds that satisfactory progress is being made.

ARTICLE 10

PROTECTION OF PERSONS AND PROPERTY

10.1 SAFETY PRECAUTIONS AND PROGRAMS

- 10.1.1 The Contractor shall be responsible for initiating, maintaining, and supervising all safety precautions and programs in connection with the performance of the contract.
- 10.1.2 To the fullest extent permitted by law, the Contractor shall defend, indemnify and hold harmless the Owner Representative, the State of Alaska, its agents and employees from and against claims, damages, losses and expenses, including, but not limited to, attorneys' fees, arising out of or resulting from the presence, uncovering or release of suspected or confirmed hazardous materials to the extent caused by the negligent acts or omissions of Contractor or the failure of Contractor to comply with the terms and conditions of the contract documents.

10.2 SAFETY OF PERSONS AND PROPERTY

- 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to:
 - 10.2.1.1 employees on the work and other persons who may be affected thereby;
 - 10.2.1.2 the work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's subcontractors of whatever tier; and
 - 10.2.1.3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.
- 10.2.2 The Contractor shall give notices and comply with applicable laws, ordinances, rules, regulations and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.
- 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and performance of the contract, reasonable safeguards for safety and protection, including posting danger signs and

other warnings against hazards, promulgating safety regulations and notifying Owner Representatives and users of adjacent sites and utilities.

- 10.2.4 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the contract documents) to property referred to in clauses 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a subcontractor of any tier, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under clauses 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner Representative or Engineer or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under paragraph 3.17.
- 10.2.5 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner Representative and Engineer.
- 10.2.6 The Contractor shall not loan or permit any part of the construction or site to be loaded so as to endanger its safety.
- 10.2.7 The Contractor shall promptly report both orally and in writing to the Owner Representative and Engineer all accidents arising out of or in connection with the work which cause death, personal injury or property damage, giving full details and statements of any witnesses.

10.3 HAZARDOUS MATERIALS

- 10.3.1 If reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated bipbyenel (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop work in the affected area and report the condition to the Owner Representative and Engineer in writing.
- 10.3.2 The Contractor shall comply with the Federal Hazard Communications standards as well as other applicable environmental laws, including, but not limited to, all laws dealing with the removal and disposal of asbestos and soil contaminated with any hazardous substance. The Contractor shall ensure that all hazardous substances with which it deals receive safe and proper handling. The Contractor shall properly perform services in connection with decontamination of construction equipment and disposal of contaminated debris or samples.

10.4 EMERGENCIES

10.4.1 In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in paragraph 4.3 and Article 7.

INSURANCE AND BONDS

11.1 CONTRACTOR'S LIABILITY INSURANCE

11.1.1 Without limiting Contractor's indemnification, it is agreed that Contractor shall purchase at its own expense and maintain in force at all times during the performance of services under this contract the following policies of insurance. Where specific limits are shown, it is understood that they shall be the minimum acceptable limits. If the Contractor's policy contains higher limits, the Owner Representative shall be entitled to coverage to the extent of such higher limits. Certificates of insurance must be furnished prior to award of contract and must provide for a 30 day prior notice of cancellation, non-renewal or material change of conditions. Failure to furnish satisfactory evidence of insurance or lapse of the policy is a material breach of this contract and shall be grounds for termination of the Contractor's services. All insurance policies shall comply with and be issued by insurers licensed to transact the business of insurance under AS.21. The Contractor shall not allow any subcontractor of any tier to commence work on any subcontract until the insurance required has been obtained. Each subcontract work to be performed will specifically include a provision that the Owner Representative, the State of Alaska, its agents and employees are not liable for damages or claims from damages arising from the subcontractor's performance or activities under the terms of the subcontracts.

11.1.2 **Workers' Compensation Insurance**: The Contractor shall provide and maintain, for all employees engaged in work under this contract, coverage as required by AS.23.30.045, and where applicable, any other statutory obligations including but not limited to Federal U.S.L. & H. and Jones Act requirements. In the case of subcontractor work, the Contractor shall provide workmen's compensation insurance for all the subcontractors' employees engaged in such work. **The policy shall provide waiver of subrogation against the State of Alaska.** The coverage shall include:

Employer's Liability Protection at \$500,000 each accident/each employee and \$500,000 policy limit

11.1.3 **Commercial General Liability Insurance**: The Contractor shall provide and maintain coverage for all business premises and operations used by the Contractor in the performance of services under this contract with minimum combined single limit coverage per the following schedule:

\$1,000,000 each occurrence \$1,000,000 personal injury \$1,000,000 general aggregate \$1,000,000 products completed operations aggregate

The State of Alaska shall be named as an additional insured.

11.1.4 **Commercial Automobile Liability Insurance**: The Contractor shall provide and maintain coverage for all vehicles used by the Contractor in the performance of services under this contract with minimum coverage limits of \$1,000,000 combined single limit per occurrence.

All of the above insurance coverage shall be considered to be primary and non-contributory to any other insurance carried by the State of Alaska, whether through self-insurance or otherwise.

11.2 GENERAL REQUIREMENTS

- 11.2.1 All insurance coverage procured by the Contractor shall be provided by insurance companies having policy holder ratings no lower than "excellent" in the Best's Insurance Guide, latest edition in effect as of the date of the contract, and subsequently in effect at the time of renewal of any policies required by the contract documents.
- 11.2.2 If the Owner Representative or the Contractor is damaged by the failure of the other party to purchase or maintain insurance required by the contract documents, then the party who failed to purchase or maintain the insurance shall bear all reasonable costs (including attorneys' fees and court and settlement expenses) attributable thereto.
- 11.2.3 <u>Certificate of Insurance</u>. Prior to commencing any work, the Contractor shall furnish certificates issued to the Owner Representative showing the type, amount, class of operations covered, effective date, and dates of expiration of such policies. Such certificates shall contain substantially the following statement: "The insurance covered by this certificate will not be canceled or materially altered except after 30 days' written notice has been received by the Owner Representative."
- 11.2.4 All certificates and policies shall contain the project title. By requiring insurance, Owner Representative makes no representation that the required coverage or limits are necessarily adequate. Such coverage and limits requirements shall not be construed as a limit on Contractor's liability or duties of indemnity.

11.3 PERFORMANCE BOND AND PAYMENT BOND

- 11.3.1 The Contractor shall furnish, prior to commencing any work, performance and payment bonds furnished by a corporate surety qualified to do business in Alaska and otherwise acceptable to the Owner Representative, each with a penal sum equaling 100% of the contract sum. The performance bond shall name as obliges the Village Safe Water Projectⁱ and any other entity as required elsewhere in the contract documents or bidding requirements.
- 11.3.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the contract, the Contractor shall promptly furnish a copy of the bonds or shall permit a copy to be made.

UNCOVERING AND CORRECTION OF WORK

12.1 UNCOVERING AND CORRECTION OF WORK

- 12.1.1 If a portion of the work is covered contrary to the Engineer's request or to requirements specifically expressed in the contract documents, it must, if required in writing by the Engineer, be uncovered for the Engineer's observation and be replaced at the Contractor's expense without change in the contract time.
- 12.1.2 If a portion of the work has been covered which the Engineer has not specifically requested to observe prior to it being covered, the Owner Representative may request to see such work and it shall be uncovered by the Contractor. If such work is in accordance with the contract documents, costs of uncovering and replacement shall, by appropriate change order, be charged to the Owner Representative. If such work is not in accordance with the contract documents, the Contractor shall pay such costs unless the condition was caused by the Owner Representative or a separate Contractor in which event the Owner Representative shall be responsible for payment of such costs.

12.2 CORRECTION OF WORK

12.2.1 The Contractor shall promptly correct work rejected by the Engineer, or failing to conform to the requirements of the contract documents, whether observed before or after substantial completion and whether or not fabricated, installed or completed. The Contractor shall bear costs of correcting such rejected work, including additional testing and inspections and compensation for the Engineer's services and expenses made necessary thereby.

12.3 AFTER SUBSTANTIAL COMPLETION

- 12.3.1 In addition to the Contractor's obligations under paragraph 3.5, if, within one year after the date of substantial completion of the work or designated portion thereof, or after the date for commencement of warranties established under subparagraph 9.7.4, or by terms of an applicable special warranty required by the contract documents, any of the work is found to be not in accordance with the requirements of the contract documents, the Contractor shall correct it promptly after receipt of written notice from the Owner Representative to do so unless the Owner Representative has previously given the Contractor a specific written acceptance of that specific condition. The Owner Representative shall give such notice promptly after discovery of the condition. If the Contractor fails to correct nonconforming work within a reasonable time, the Owner Representative may correct it in accordance with paragraph 2.4.
 - 12.3.1.1 This period of one year shall be extended with respect to portions of work first performed after substantial completion by the period of time between substantial completion and the actual performance of the work.
 - 12.3.1.2 Upon completion of any repairs or replacement pursuant to this paragraph, the one year correction period in connection with the work requiring corrections shall be renewed and recommence.

- 12.3.1.3 If the Contractor does not proceed with correction of such nonconforming work within a reasonable time, the Owner Representative may remove it and store the salvable materials or equipment at the Contractor's expense. If the Contractor does not pay costs of such removal and storage within 10 days after written notice, the Owner Representative may upon 10 additional days' written notice sell such materials and equipment at auction or at private sale and shall account for the proceeds thereof, after deducting costs and damages that should have been borne by the Contractor, including compensation for the Engineer's services and expenses made necessary thereby. The contract sum shall be reduced by any deficiency. If payments then or thereafter due the Contractor are not sufficient to cover such amount, the Contractor shall pay the difference to the Owner Representative.
- 12.3.2 The Contractor shall remove from the site portions of the work which are not in accordance with the requirements of the contract documents and are neither corrected by the Contractor nor accepted by the Owner Representative.
- 12.3.3 The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner Representative or separate Contractors caused by the Contractor's correction or removal of work which is not in accordance with the requirements of the contract documents.
- 12.3.4 Nothing contained in paragraph 12.2 shall be construed to establish a period of limitation with respect to other obligations which the Contractor might have under the contract documents. Establishment of the time period of one year as described in subparagraph 12.3 relates only to the specific obligation of the Contractor to correct the work, and has no relationship to the time within which the obligation to comply with the contract documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the work.
- 12.3.5 This obligation under this subparagraph 12.3 shall survive acceptance of the work under the contract and termination of the contract.
- 12.3.6 The obligations of paragraph 12.2 shall cover any repairs or replacement to any part of the work or other property caused by the defective work.

12.4 ACCEPTANCE OF NONCONFORMING WORK

12.4.1 If the Owner Representative prefers to accept work which is not in accordance with the requirements of the contract documents, the Owner Representative may do so instead of requiring its removal and correction, in which case the contract sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

MISCELLANEOUS PROVISIONS

13.1 GOVERNING LAW

13.1.1 The contract shall be governed under the State of Alaska's Law.

13.2 SUCCESSORS AND ASSIGNS

13.2.1 The Owner Representative and Contractor respectively bind themselves, their partners, successors, assign and legal representatives to the other party hereto and to partners, successors, assign and legal representatives of such other party in respect to covenants, agreements and obligations contained in the contract documents. Neither party to the contract shall assign the contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the contract.

13.3 WRITTEN NOTICE

13.3.1 Written notice shall be deemed to have been duly served if delivered in person to the individual, or a member of the firm or entity, or to an officer of the corporation for which it was intended, or if delivered at or sent by registered or certified mail to the last business address known to the party giving notice.

13.4 RIGHTS AND REMEDIES

- 13.4.1 Duties and obligations imposed by the contract documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.
- 13.4.2 No action or failure to act by the Owner Representative, Engineer or Contractor shall constitute a waiver of a right or duty afforded them under the contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach thereunder, except as may be specifically agreed in writing.
- 13.4.3 Owner Representative shall have until six years from the date damage to any improvement to real property constructed pursuant to this agreement is actually discovered by Owner Representative in which to bring any claim related to such damage against any person who may be liable to Owner Representative in whole or in part for such damage including, but not limited to, Contractor. Any shorter period in which to make such a claim imposed by AS 09.10.055 is expressly excluded from this agreement.

13.5 TESTS AND INSPECTIONS

13.5.1 Tests, inspections and approvals of portions of the work required by the contract documents or by laws, ordinances, rules, regulations or orders of public authorities having jurisdiction shall be made at an appropriate time. Unless otherwise provided, the Contractor shall make arrangements for such

tests, inspections, and approvals with an independent testing laboratory or entity acceptable to the Owner Representative, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Engineer timely notice of when and where tests and inspections are to be made so the Engineer may observe such procedures. The Owner Representative shall bear costs of tests, inspections, or approvals which do not become requirements until after bids are received or negotiations concluded, and not required because of a previously failed test or defect in the work.

- 13.5.2 If the Engineer, Owner Representative or public authorities having jurisdiction determine that portions of the work require additional testing, inspection, or approval not included under subparagraph 13.5.1, the Engineer shall, upon written authorization from the Owner Representative, instruct the Contractor to make arrangements for such additional testing, inspection, or approval by an entity acceptable to the Owner Representative, and the Contractor shall give timely notice to the Engineer of when and where tests and inspections are to be made so the Engineer may observe such procedures. The Owner Representative shall bear such costs except as provided in subparagraph 13.5.3.
- 13.5.3 If such procedures for testing, inspection, or approval under subparagraphs 13.5.1 and 13.5.2 reveal failure of the portions of the work to comply with requirements established by the contract documents, the Contractor shall bear all costs made necessary by such failure including those of repeated procedures and compensation for the Engineer's services and expenses.
- 13.5.4 Required certificates of testing, inspection, or approval shall, unless otherwise required by the contract documents, be secured by the Contractor and promptly delivered to the Engineer.
- 13.5.5 If the Engineer is to observe tests, inspections, or approvals required by the contract documents, the Engineer shall do so promptly and, where practicable, at the normal place of testing.
- 13.5.6 Tests or inspections conducted pursuant to the contract documents shall be made promptly to avoid unreasonable delay in the work.

13.6 GENERAL PROVISIONS

- 13.6.1 Whenever possible, each provision of this agreement shall be interpreted in a manner so as to be effectively valid under applicable law. If, however, any provision of this agreement, or portion thereof, is prohibited by law or found invalid under any law, only such provision or portion thereof shall be ineffective, without in any manner invalidating or affecting the remaining provisions of this agreement or valid portions of such provision, which are hereby deemed severable.
- 13.6.2 Any specific requirement in this contract that the responsibility or obligations of the Contractor shall also apply to a subcontractor is added for emphasis and are also hereby deemed to include a subcontractor of any tier. The omission of a reference to a subcontractor of any tier in connection with any of the Contractor's responsibilities or obligations shall not be construed to diminish, abrogate or limit any responsibilities or obligations of a subcontractor of any tier under the contract documents or the applicable subcontract.

TERMINATION OR SUSPENSION OF THE CONTRACT

14.1 TERMINATION BY THE OWNER REPRESENTATIVE FOR CAUSE

- 14.1.1 The Owner Representative may terminate the contract if the Contractor:
 - 14.1.1.1 Persistently or repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
 - 14.1.1.2 Fails to make payment to subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the subcontractors;
 - 14.1.1.3 Disregards laws, ordinances, or rules, regulations or orders of a public authority having jurisdiction;
 - 14.1.1.4 Otherwise is guilty of substantial breach of a provision of the contract documents;
 - 14.1.1.5 Breaches a warranty made by the Contractor under or pursuant to the contract documents; or
 - 14.1.1.6 Fails after commencement of the work to proceed continuously with the construction and completion of the work for more than 10 days except as permitted under the contract documents.
- 14.1.2 When any of the above reasons exist, the Owner Representative may without prejudice to any other rights or remedies of the Owner Representative and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:
 - 14.1.2.1 Take possession of the site and of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
 - 14.1.2.2 Accept assignment of subcontracts pursuant to paragraph 5.4; and
 - 14.1.2.3 Finish the work by whatever reasonable method the Owner Representative may deem expedient.
- 14.1.3 When the Owner Representative terminates the contract for one of the reasons stated in subparagraph 14.2.1, the Contractor shall not be entitled to receive further payment until the work is finished.
- 14.1.4 If the unpaid balance of the contract sum exceeds costs of finishing the work, including compensation for the Engineer's services and expenses made necessary thereby, such excess shall be paid to the Contractor. If such costs exceed the unpaid balance, the Contractor shall pay the difference

to the Owner Representative. The amount to be paid to the Contractor or Owner Representative, as the case may be, shall be certified by the Engineer, upon application, and this obligation for payment shall survive termination of the contract.

14.3 SUSPENSION BY THE OWNER REPRESENTATIVE FOR CONVENIENCE

14.3.1 The Owner Representative may, without cause, order a Contractor in writing to suspend, delay or interrupt the work in whole or in part for such period of time as the Owner Representative may determine.

14.4 TERMINATION BY THE OWNER REPRESENTATIVE FOR CONVENIENCE

- 14.4.1 The Owner Representative may, at any time, terminate the contract for the Owner Representative's convenience and without cause.
- 14.4.2 Upon receipt of written notice from the Owner Representative of such termination for the Owner Representative's convenience, the Contractor shall:
 - 14.4.2.1 Cease operations as directed by the Owner Representative in the notice;
 - 14.4.2.2 Take actions necessary, or that the Owner Representative may direct, for the protection and preservation of the work; and
 - 14.4.2.3 Except for work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

In case of such termination for the Owner Representative's convenience, the Contractor shall be entitled to receive payment for work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the work not executed.

14.5 Effect of Termination

Upon termination by Owner Representative, the Contractor shall: (1) Stop work as directed by Owner Representative. Place no further orders or requests of subcontractors, if any, for goods or services; (2) Take actions necessary, or that Owner Representative may direct, for the protection and preservation of the goods or services; (3) Terminate all orders and subcontracts to the extent that they relate to the performance of work terminated by the termination notice; (4) Deliver or otherwise make available to Owner Representative all data, reports, estimates, confidential information, summaries and such other information and materials, as may have been accumulated by the Contractor in performing the contract, whether completed or in process.

END OF GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

i Reference Project Name and number

Federal Debarment Certification Form

Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion Lower Tier Covered Transactions

This certification is required by the regulations implementing Executive Order 12549, Debarment and Suspension, 29 CFR Part 98, Section 98.510, Participant's responsibilities. The regulations were published as Part VII of the May 26, 1988 Federal Register (pages 19160-19211).

(BEFORE COMPLETING CERTIFICATION, READ THE INSTRUCTIONS ON THE FOLLOWING PAGE WHICH ARE AN INTEGRAL PART OF THE CERTIFICATION)

- (1) The prospective recipient of Federal assistance funds certifies, by submission of this bid, that neither it nor its principals are presently debarred, suspended, proposed for debarment, declared ineligible, or voluntarily excluded from participation in this transaction by any Federal department or agency.
- (2) Where the prospective recipient of Federal assistance funds is unable to certify to any of the Statements in this certification, such prospective participant shall attach an explanation to this Proposal.

Name and Title of Authorized Representative		
Signature	Date	

Federal Debarment Certification Form Instructions

Instructions for Certification

- 1. By signing and submitting this Proposal, the prospective recipient of Federal assistance funds is providing the certification as set out below.
- 2. The certification in this class is a material representation of fact upon which reliance was placed when this transaction was entered into. If it is later determined that the prospective recipient of Federal assistance funds knowingly rendered an erroneous certification, in addition to other remedies available to the Federal Government, the Department of Labor (DOL) may pursue available remedies, including suspension and/or debarment.
- 3. The prospective recipient of Federal assistance funds shall provide immediate written notice to the person to whom this Proposal is submitted if at any time the prospective recipient of Federal assistance funds learns that its certification was erroneous when submitted or has become erroneous by reason of changed circumstances.
- 4. The terms "covered transaction," "debarred," "suspended," "ineligible," "lower tier covered transaction," "participant," "person," "primary covered transaction," "principal," "Proposal," and "voluntarily excluded," as used in this clause, have the meanings set out in the Definitions and Coverage sections of rules implementing Executive Order 12549. You may contact the person to which this Proposal is submitted for assistance in obtaining a copy of those regulations.
- 5. The prospective recipient of Federal assistance funds agrees by submitting this Proposal that, should the proposed covered transaction be entered into, it shall not knowingly enter into any lower tier covered transaction with a person who is debarred, suspended, declared ineligible, or voluntarily excluded from participation in this covered transaction, unless authorized by the DOL.
- 6. The prospective recipient of Federal assistance funds further agrees by submitting this Proposal that it will include the clause titled "Certification Regarding Debarment, Suspension, Ineligibility and Voluntary Exclusion Lower Tier Covered Transactions," without modification, in all lower tier covered transactions and in all solicitations for lower tier covered transactions.
- 7. A participant in a covered transaction may rely upon a certification of a prospective participant in a lower tier covered transaction that it is not debarred, suspended, ineligible, or voluntarily excluded from the covered transaction, unless it knows that the certification is erroneous. A participant may decide the method and frequency by which it determines the eligibility of its principals. Each participant may but is not required to check the List of Parties Excluded from Procurement or Non-procurement Programs.
- 8. Nothing contained in the foregoing shall be construed to require establishment of a system of records in order to render in good faith the certification required by this clause. The knowledge and information of a participant is not required to exceed that which is normally possessed by a prudent person in the ordinary course of business dealings.
- 9. Except for transactions authorized under paragraph 5 of these instructions, if a participant in a covered transaction knowingly enters into a lower tier covered transaction with a person who is suspended, debarred, ineligible, or voluntarily excluded from participation in this transaction, in addition to other remedies available to the Federal Government, the DOL may pursue available remedies, including suspension and/or debarment.

FAR 52.203-11

CERTIFICATION AND DISCLOSURE REGARDING PAYMENTS TO INFLUENCE CERTAIN FEDERAL TRANSACTIONS (SEPT 2007) (a) Definitions. As used in this

provision—"Lobbying contact" has the meaning provided at 2

<u>U.S.C. 1602(8)</u>. The terms "agency," "influencing or attempting to influence," "officer or employee of an agency," "person," "reasonable compensation," and "regularly employed" are defined in the FAR clause of this solicitation entitled "Limitation on Payments to Influence Certain Federal Transactions" (52.203-12).

- (b) *Prohibition*. The prohibition and exceptions contained in the FAR clause of this solicitation entitled "Limitation on Payments to Influence Certain Federal Transactions" (52.203-12) are hereby incorporated by reference in this provision.
- (c) Certification. The offeror, by signing its offer, hereby certifies to the best of its knowledge and belief that no Federal appropriated funds have been paid or will be paid to any person for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress on its behalf in connection with the awarding of this contract.
- (d) *Disclosure*. If any registrants under the Lobbying Disclosure Act of 1995 have made a lobbying contact on behalf of the offeror with respect to this contract, the offeror shall complete and submit, with its offer, OMB Standard Form LLL, Disclosure of Lobbying Activities, to provide the name of the registrants. The offeror need not report regularly employed officers or employees of the offeror to whom payments of reasonable compensation were made.
- (e) *Penalty*. Submission of this certification and disclosure is a prerequisite for making or entering into this contract imposed by <u>31 U.S.C. 1352</u>. Any person who makes an expenditure prohibited under this provision or who fails to file or amend the disclosure required to be filed or amended by this provision, shall be subject to a civil penalty of not less than \$10,000, and not more than \$100,000, for each such failure.

CONSENT TO USE OF ELECTRONIC SIGNATURES

BY CHECKING HERE, I AGREE TO THE USE OF ELECTRONIC SIGNATURES AS VALID, LEGALLY BINDING SUBSTITUTES FOR ORIGINAL, HANDWRITTEN SIGNATURES ON THIS DOCUMENT.

Company	
Name (signature)	
Name (printed)	
Title	Date of execution

APPENDIX C Page 1 of 3

DISCLOSURE OF LOBBYING ACTIVITIES

Approved by OMB 0348-0046

Authorized for Local Reproduction

Standard Form LLL (Rev. 7-97)

Complete this form to disclose lobbying activities pursuant to 31 U.S.C. 1352

8. Federal Action Number, if known:

Federal Use Only:

10. a. Name and Address of Lobbying Registrant

(if individual, last name, first name, MI):

(See reverse for public burden disclosure.) 1. Type of Federal Action: 2. Status of Federal Action: 3. Report Type: a. contract a. bid/offer/application a. initial filing b. grant [⊥]b. initial award b. material change c. cooperative agreement c. post-award For Material Change Only: year _____ quarter ____ e. loan guarantee date of last report f. loan insurance 4. Name and Address of Reporting Entity: 5. If Reporting Entity in No. 4 is a Subawardee, Enter Name Subawardee and Address of Prime: Prime Tier _____, if known: Congressional District, if known: **Congressional District**, *if known*: 6. Federal Department/Agency: 7. Federal Program Name/Description:

CFDA Number, if applicable: _____

b. Individuals Performing Services (including address if

9. Award Amount, if known:

different from No. 10a) (last name, first name, MI):

11. Information requested through this form is authorized by title 31 U.S.C. section 1352. This disclosure of lobbying activities is a material representation of fact Signature: upon which reliance was placed by the tier above when this transaction was made Print Name: _____ or entered into. This disclosure is required pursuant to 31 U.S.C. 1352. This information will be available for public inspection. Any person who fails to file the required disclosure shall be subject to a civil penalty of not less than \$10,000 and not more than \$100,000 for each such failure. Telephone No.: _____ _____ Date: ____

APPENDIX C Page 2 of 3

INSTRUCTIONS FOR COMPLETION OF SF-LLL, DISCLOSURE OF LOBBYING ACTIVITIES

This disclosure form shall be completed by the reporting entity, whether subawardee or prime Federal recipient, at the initiation or receipt of a covered Federal action, or a material change to a previous filing, pursuant to title 31 U.S.C. section 1352. The filing of a form is required for each payment or agreement to make payment to any lobbying entity for influencing or attempting to influence an officer or employee of any agency, a Member of Congress, an officer or employee of Congress, or an employee of a Member of Congress in connection with a covered Federal action. Complete all items that apply for both the initial filing and material change report. Refer to the implementing guidance published by the Office of Management and Budget for additional information.

- 1. Identify the type of covered Federal action for which lobbying activity is and/or has been secured to influence the outcome of a covered Federal action.
- 2. Identify the status of the covered Federal action.
- 3. Identify the appropriate classification of this report. If this is a followup report caused by a material change to the information previously reported, enter the year and quarter in which the change occurred. Enter the date of the last previously submitted report by this reporting entity for this covered Federal action.
- 4. Enter the full name, address, city, State and zip code of the reporting entity. Include Congressional District, if known. Check the appropriate classification of the reporting entity that designates if it is, or expects to be, a prime or subaward recipient. Identify the tier of the subawardee, e.g., the first subawardee of the prime is the 1st tier. Subawards include but are not limited to subcontracts, subgrants and contract awards under grants.
- 5. If the organization filing the report in item 4 checks "Subawardee," then enter the full name, address, city, State and zip code of the prime Federal recipient. Include Congressional District, if known.
- 6. Enter the name of the Federal agency making the award or loan commitment. Include at least one organizationallevel below agency name, if known. For example, Department of Transportation, United States Coast Guard.
- 7. Enter the Federal program name or description for the covered Federal action (item 1). If known, enter the full Catalog of Federal Domestic Assistance (CFDA) number for grants, cooperative agreements, loans, and loan commitments.
- 8. Enter the most appropriate Federal identifying number available for the Federal action identified in item 1 (e.g., Request for Proposal (RFP) number; Invitation for Bid (IFB) number; grant announcement number; the contract, grant, or loan award number; the application/proposal control number assigned by the Federal agency). Include prefixes, e.g., "RFP-DE-90-001."
- 9. For a covered Federal action where there has been an award or loan commitment by the Federal agency, enter the Federal amount of the award/loan commitment for the prime entity identified in item 4 or 5.
- 10. (a) Enter the full name, address, city, State and zip code of the lobbying registrant under the Lobbying Disclosure Act of 1995 engaged by the reporting entity identified in item 4 to influence the covered Federal action.
 - (b) Enter the full names of the individual(s) performing services, and include full address if different from 10 (a). Enter Last Name, First Name, and Middle Initial (MI).
- 11. The certifying official shall sign and date the form, print his/her name, title, and telephone number.

According to the Paperwork Reduction Act, as amended, no persons are required to respond to a collection of information unless it displays a valid OMB Control Number. The valid OMB control number for this information collection is OMB No. 0348-0046. Public reporting burden for this collection of information is estimated to average 10 minutes per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding the burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to the Office of Management and Budget, Paperwork Reduction Project (0348-0046), Washington, DC 20503.

APPENDIX C Page 3 of 3

STATE OF ALASKA DEPARTMENT OF ENVIRONMENTAL CONSERVATION Village Safe Water Program

SUBCONTRACTOR LIST

	Project Na	me and Number			
The apparent low bidder shall complete this form and submit it so as to be received by the Procurement Officer prior to the close of business on the fifth working day after receipt of written notice from the Department of Environmental Conservation.					
An apparent low bidder who fails to submit a completed Subcontractor List form within the time allowed will be declared nonresponsible and may be required to forfeit the bid security.					
Scope of work must be clearly defined. If an item of work is to be performed by more than one firm, indicate the portion or percent of work to be done by each.					
Check as applicable: [] All Work on the above-referenced project will be accomplished without subcontracts Or					
[□] List a	ll first tier Subcor	ntractors as follows:			
FIRM NAME, ADDRESS, PHONE NO.	CONTR	S LICENSE NO., RACTOR'S RATION NO.	SCOPE OF WORK TO BE PERFORMED		
CONTINU	JE SUBCONTRAC	TOR INFORMATION	ON REVERSE		
For projects with federal-aid funding, I hereby certify Alaska Business Licenses and Contractor Registrations will be valid for all subcontractors prior to award of the subcontract. For projects without federal-aid funding (State funding only), I hereby certify the listed Alaska Business Licenses and Contractor Registrations were valid at the time bids were opened for this project.					
Signature of Authorized Company Representative Title					
Company Name		Company Addres	ss (Street or PO Box, City, State, Zip)		
Date		Phone Number			

Appendix D

FIRM NAME, ADDRESS, PHONE NO.	AK BUSINESS LICENSE NO., CONTRACTOR'S REGISTRATION NO.	SCOPE OF WORK TO BE PERFORMED